UNIVERSITY OF BIRMINGHAM University of Birmingham Research at Birmingham

Timing and Consequences of Bering Strait Opening

Hall, Jonathan; Allison, Matthew; Papadopoulos, Max; Barfod, Dan N.; Jones, Stephen

DOI: 10.1029/2022PA004539

License: Creative Commons: Attribution (CC BY)

Document Version Publisher's PDF, also known as Version of record

Citation for published version (Harvard):

Hall, J, Allison, M, Papadopoulos, M, Barfod, DN & Jones, S 2023, 'Timing and Consequences of Bering Strait Opening: New Insights From ⁴ Ar/³ Ar Dating of the Barmur Group (Tjörnes Beds), Northern Iceland', *Paleoceanography and Paleoclimatology*, vol. 38, no. 4, e2022PA004539. https://doi.org/10.1029/2022PA004539

Link to publication on Research at Birmingham portal

General rights

Unless a licence is specified above, all rights (including copyright and moral rights) in this document are retained by the authors and/or the copyright holders. The express permission of the copyright holder must be obtained for any use of this material other than for purposes permitted by law.

•Users may freely distribute the URL that is used to identify this publication.

•Users may download and/or print one copy of the publication from the University of Birmingham research portal for the purpose of private study or non-commercial research. •User may use extracts from the document in line with the concept of 'fair dealing' under the Copyright, Designs and Patents Act 1988 (?)

•Users may not further distribute the material nor use it for the purposes of commercial gain.

Where a licence is displayed above, please note the terms and conditions of the licence govern your use of this document.

When citing, please reference the published version.

Take down policy

While the University of Birmingham exercises care and attention in making items available there are rare occasions when an item has been uploaded in error or has been deemed to be commercially or otherwise sensitive.

If you believe that this is the case for this document, please contact UBIRA@lists.bham.ac.uk providing details and we will remove access to the work immediately and investigate.



Paleoceanography and Paleoclimatology^{*}

RESEARCH ARTICLE

10.1029/2022PA004539

Key Points:

- New Ar-Ar dating shows a late Miocene to early Pliocene (6.0–4.4 Ma) age range for the Barmur Group (Tjörnes beds), northern Iceland
- Arrival of Pacific bivalves in Iceland is dated at 5.2–4.9 Ma (chrons C3n.4n–C3n.3r), soon after first Bering Strait opening at 5.6–5.4 Ma
- Opening of the Bering Strait gateway did not directly cause the development of large northern hemisphere icecaps

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

J. R. Hall and S. M. Jones, j.r.hall@bham.ac.uk; s.jones.4@bham.ac.uk

Citation:

Hall, J. R., Allison, M. S., Papadopoulos, M. T., Barfod, D. N., & Jones, S. M. (2023). Timing and consequences of Bering Strait opening: New insights from ⁴⁰Ar/³⁹Ar dating of the Barmur Group (Tjörnes beds), northern Iceland. *Paleoceanography and Paleoclimatology*, 38, e2022PA004539 https://doi. org/10.1029/2022PA004539

Received 5 SEP 2022 Accepted 27 MAR 2023

Author Contributions:

Conceptualization: Jonathan R. Hall, Stephen M. Jones Formal analysis: Jonathan R. Hall, Dan N. Barfod, Stephen M. Jones Investigation: Jonathan R. Hall, Matthew S. Allison, Max T. Papadopoulos Methodology: Jonathan R. Hall, Matthew S. Allison, Max T. Papadopoulos Supervision: Stephen M. Jones Visualization: Jonathan R. Hall Writing – original draft: Jonathan R. Hall Writing – review & editing: Matthew S. Allison, Dan N. Barfod, Stephen M. Jones

© 2023. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Timing and Consequences of Bering Strait Opening: New Insights From ⁴⁰Ar/³⁹Ar Dating of the Barmur Group (Tjörnes Beds), Northern Iceland

Jonathan R. Hall¹, Matthew S. Allison¹, Max T. Papadopoulos¹, Dan N. Barfod², and Stephen M. Jones¹

¹School of Geography, Earth and Environmental Sciences, University of Birmingham, Birmingham, UK, ²NERC Argon Isotope Facility, Scottish Universities Environmental Research Centre, East Kilbride, UK

Abstract The Barmur Group (informally Tjörnes beds) sedimentary succession of northern Iceland is key to reconstructing the opening of the Bering Strait oceanic gateway because these rocks record migration of bivalve molluscs from the Pacific to the Atlantic via the Arctic. However, the timing of the migration event is poorly constrained owing to a lack of reliable absolute ages. To address this problem, we present the first Ar-Ar radiometric dates from four basaltic lavas that underlie, are intercalated with, and overlie the Barmur Group, and integrate them with existing paleomagnetic records. We show that the Barmur Group has a latest Miocene to early Pliocene age range (c. 6.0-4.4 Ma; C3r-C3n.2n), older than all previous age models. Thus, the Barmur Group does not record the mid-Piacenzian Warm Period, contra some previous suggestions. Abundant Pacific bivalve molluscs appeared in the Barmur Group during subchrons C3n.4n-C3n.3r at 5.235-4.896 Ma, over 1.3 million years earlier than previously suggested. Appearance of Pacific bivalves in the northern Atlantic occurred shortly after the 5.6–5.4 Ma age previously inferred for first appearance of Arctic bivalves in the Pacific. Thus, our data suggest that first opening of the Bering Strait gateway by the latest Miocene (c. 5.5 Ma) was soon followed by bidirectional trans-Arctic faunal exchange, and argue against a hypothesized two-stage faunal exchange process spanning c. 2 million years. Our results also confirm that first opening of the Bering Strait gateway was not directly associated with the growth of large northern hemisphere icesheets, which occurred several million years later.

Plain Language Summary The extent to which global climate change is influenced by the flow of seawater between the world's oceans is an important question in Earth science. Sedimentary rocks from the Tjörnes Peninsula in northern Iceland record the opening of the Bering Strait oceanic gateway, which connects the Pacific and Atlantic Oceans via the Arctic. The signature of this gateway opening event is the sudden appearance in Iceland of bivalve mollusc fossils that previously lived only in the Pacific Ocean. However, the timing of their arrival is uncertain owing to the lack of reliable age measurements. To solve this problem, we applied radioisotope dating methods to a succession of rocks called the Barmur Group from the Tjörnes Peninsula. We found that the sudden appearance of Pacific fossils occurred over a million years earlier than previously thought. Our data show that first opening of the Bering Strait gateway was soon followed by migration of marine faunas in both directions, in contrast to the existing hypothesis that Pacific-to-Atlantic migration significantly post-dated Atlantic-to-Pacific migration. Our results also confirm the hypothesis that establishment of the Pacific-Arctic-Atlantic oceanic connection did not directly trigger development of large northern hemisphere icecaps, which occurred several million years later.

1. Introduction

The Barmur Group sedimentary succession in north-eastern Iceland, informally known as the Tjörnes beds, is an important archive of late Neogene palaeoenvironmental change in the high-latitude North Atlantic (Figure 1a, Símonarson & Eiríksson, 2021a). The easily accessible, c. 600 m thick succession contains a variety of macro-fossils and microfossils that have been studied for a century (Bárðarson, 1925; Eiríksson & Símonarson, 2021). Notably, the Barmur Group is a key outcrop for dating the first opening of the Bering Strait oceanic gateway, which established a trans-Arctic marine connection between the Pacific and Atlantic Oceans (Marincovich & Gladenkov, 1999; Símonarson & Eiríksson, 2021a). Opening of the Bering Strait allowed relatively fresh Pacific water into the Arctic and North Atlantic, which may have reduced the strength of Atlantic Meridional







Figure 1. Location and geological setting of the Barmur Group. (a) Geographical location of the Tjörnes succession, northern Iceland. Field area highlighted by yellow box. (b) Geological map of the Tjörnes succession modified after Eiríksson (1980), using the geocentric coordinate system for Iceland (ISN2004/Lambert 2004; EPSG:5325). Place names discussed in the text are shown. Topographic basemap created in QGIS v.3.22 using the Landmælingum Íslands (National Land Survey of Iceland) Hæðarlíkan 2016 Digital Elevation Model. (c) Composite lithostratigraphic column of the Barmur Group after Eiríksson et al. (2021b), where width increases with grain size. The subdivision of the Barmur Group into formations and members is shown, alongside the molluscan biozonation, marine (Beds 1–25) and terrestrial units (Beds A–J) of Bárðarson (1925). Sample locations and abbreviations are indicated by yellow stars.





Figure 2. Comparison of existing age models from the Barmur Group with our new age model. Our new, older age for the base of the Höskuldsvík Group could be accommodated in several possible ways (options 1–3). The units are correlated with the Geomagnetic Polarity Time Scale of Ogg (2020). The *MactralSerripes* biozone boundary, which marks the sudden arrival of abundant Pacific molluscs to northern Iceland, is highlighted orange for contrast. Global climate, oceanic gateways and Icelandic rifting events are also shown (Doré & Lundin, 1996; Garcia et al., 2003; Jones et al., 2002; Raffi et al., 2020; Straume et al., 2020; Walters et al., 2013). V-shaped ridges in the bathymetry surrounding Iceland result from thickness variations caused by temporal fluctuations in melt production rate at the Mid-Atlantic Ridge (Jones et al., 2002). ⁴⁰Ar/³⁹Ar composite ages from lavas as follows: Ly, Lynghöfði; Gr, Grænhöfði; Sk, Skeifá; and Hö, Höskuldsvík. MSC, Messinian salinity crisis; MIS M2, Marine Isotope Stage M2; mPWP, mid-Piacenzian Warm Period; and iNHG, intensification of the Northern Hemisphere Glaciation.

Overturning Circulation (AMOC), lowered high-latitude Atlantic surface temperatures, and promoted growth of northern hemisphere icesheets (Brierley & Fedorov, 2016; Hu et al., 2010). Previous studies have suggested that the Barmur Group may also record the mid-Piacenzian Warm Period (mPWP) (Cronin, 1991; Dowsett & Poore, 1991), the most recent geological time slice (3.264–3.025 Ma) when atmospheric carbon dioxide concentrations were similar to present, and surface temperatures were as predicted for the coming century (Dowsett et al., 2016). However, uncertainty over the age of the Barmur Group means that its potential to shed light on these significant changes in global oceanic circulation and climate has not been fully realized (Figure 2).

Reconstruction of the first opening of the Bering Strait oceanic gateway is based primarily on a handful of outcrops bordering the northern Pacific and northern Atlantic Oceans that contain marine bivalve molluscs. Mollusc assemblages in the Pacific and Atlantic realms were distinct during the middle to late Miocene, indicating that the Bering Strait gateway did not yet exist and a land-bridge connected Asia with North America (Marincovich, 2000; Marincovich & Gladenkov, 1999, 2001). The first post-Cretaceous opening of an oceanic gateway is signaled by the appearance of Atlantic molluscs in the Pacific, and vice versa (Gladenkov & Gladenkov, 2004; Gladenkov et al., 2002). The Barmur Group contains a plentiful and well-studied molluscan fauna (Durham & MacNeil, 1967; Símonarson & Eiríksson, 2021b; Strauch, 1963, 1972a, 1972b). The arrival of abundant Pacific bivalve molluscs defines the boundary between the *Mactra* and *Serripes* biozones (Figure 2). However, the bivalves themselves cannot be used for dating due to their wide biostratigraphic ranges (Marincovich & Gladenkov, 1999). Consequently, a fundamental problem which has dogged Bering Strait gateway reconstructions is that the faunal migration signals must be dated by other means. Dating of bivalve-bearing successions in the Pacific Ocean near the gateway, using marine diatom microfossils, has established a latest possible age range of 5.6–5.4 Ma for initial opening of the Bering Strait (Gladenkov & Gladenkov, 2004; Gladenkov et al., 2002).

The Barmur Group is potentially of great importance in Bering Strait reconstructions because the arrival of the Pacific molluscs occurs just below the Skeifá basaltic lavas, which should be dateable. But despite attempts by Albertsson (1976), no study has yet yielded a reliable radiometric age for the Skeifá lavas. Indeed, the variety of age models in Figure 2 reflects the lack of a reliable radiometric age framework for the entire Barmur Group. Age estimates range from c. 3.0 to 4.2 Ma (early to late Pliocene) for the top of the succession, and from c. 4.1 to 5.9 Ma (early Pliocene to latest Miocene) for the base of the succession. It is thus unclear whether the Barmur Group actually records the mPWP, and consequently it has been deleted from newer versions of the Pliocene Research, Interpretation and Synoptic Mapping (PRISM) global palaeoenvironmental reconstruction (Dowsett et al., 2016; Robinson et al., 2018). Most of the age models in Figure 2 are based on K-Ar dates for basaltic lavas that overlie and underlie the Barmur Group (Albertsson, 1976, 1978; Aronson & Sæmundsson, 1975). However, 1970s-vintage K-Ar ages for Neogene basaltic lavas are often unrealistically young (McDougall & Harrison, 1999). The fundamental underlying problems of low initial potassium content and hydrothermal alteration that leads to argon loss are both reported to affect the Tjörnes basalt samples (Albertsson, 1976, 1978; Aronson & Sæmundsson, 1975). Furthermore, the only Barmur Group age model that is based on biostratigraphic correlation (marine dinoflagellates; Verhoeven et al., 2011) is significantly older than the radiometric age models, in line with the possibility that the vintage K-Ar ages could be too young.

Despite these uncertainties in the age model for the Barmur Group, a late Pliocene or uppermost early Pliocene age (c. 3.4-4.5 Ma, Figure 2) has become established in the literature for the arrival of Pacific bivalves in Iceland (Albertsson, 1978; Buchardt & Simonarson, 2003; Einarsson et al., 1967; Eiríksson et al., 2021a; Símonarson & Eiríksson, 2008; Verhoeven et al., 2011). This age range is distinctly younger than the arrival of the Arctic and Atlantic bivalves in the Pacific, at 5.6–5.4 Ma (Gladenkov & Gladenkov, 2004; Gladenkov et al., 2002). The distinction between these two age estimates for faunal migration has led to a two-stage model for Bering Strait opening and the consequent development of oceanic flow (Marincovich, 2000). In the first stage, at around 5.5 Ma, the gateway opened but persistent Arctic-to-Pacific flow (southward, opposite to present) meant that although Arctic and Atlantic bivalve species immediately migrated to the Pacific, the Pacific bivalve species did not migrate to the Atlantic. In the second stage, net oceanic flow through the gateway changed to northward, like the present, which allowed Pacific bivalve species to migrate across the Arctic Ocean and into the Atlantic. The younger age estimates for the arrival of Pacific bivalves in Iceland (c. 3.4 Ma) are only shortly before the intensification of the Northern Hemisphere Glaciation (iNHG), which locally began with expansion of the Greenland icesheet from the southern uplands across northern Greenland (c. 3.5-3.0 Ma, Maslin et al., 1998) and culminated in establishment of cycles of lowland glaciation in Iceland (c. 2.7 Ma, Eiríksson, Guðmundsson, et al., 2021; Geirsdóttir & Eiríksson, 1994). Climate modeling studies show that opening of the Bering Strait can potentially promote growth of northern hemisphere icesheets by reducing high-latitude Atlantic surface temperatures, which has raised the possibility of a causal connection between gateway opening and the iNHG (Brierley & Fedorov, 2016). Alternatively, the iNHG could have been a tipping-point response to either a gradual reduction in atmospheric greenhouse gases (Bartoli et al., 2011; Lunt et al., 2008; Seki et al., 2010; Willeit et al., 2015) or changes in the AMOC associated with the closure of the Central American Seaway (De Schepper et al., 2013; Driscoll & Haug, 1998; Sarnthein et al., 2009).

Here, we address the uncertainty in age of the Pacific-to-Atlantic bivalve mollusc migration event, and the consequent uncertainty in Bering Strait gateway opening, by applying the Ar-Ar dating technique to four basaltic lavas below, within and above the Barmur Group. These are the first Ar-Ar ages for the pre-Quaternary Tjörnes succession, and they include the first reliable radiometric age for the Skeifá lavas that are closely associated with the bivalve mollusc migration event. We integrate the radiometric ages with existing paleomagnetic data to generate a new age model for the entire Barmur Group and surrounding rocks, in order to reassess the paleoclimatic implications of the first opening of the Bering Strait.

2. Geological Background

A c. 1,200 m thick sequence of latest Miocene or early Pliocene to Late Pleistocene sedimentary rocks known as the Tjörnes succession is exposed as coastal cliffs and valleys along the west side of the Tjörnes Peninsula, north-eastern Iceland (Figures 1a-1c). It is the thickest sedimentary succession of onshore Iceland, and has been intensively studied owing to its accessibility and the palaeoenvironmental change it records. The Tjörnes succession is divided stratigraphically into four units. The Kaldakvísl Group of plateau tholeiite basaltic lavas forms the basement. The Barmur Group, the main focus of this study, comprises lignites, siltstones, sandstones, conglomerates, and occasional basaltic lavas which are interpreted to have accumulated in an alternating shallow

marine to terrestrial environment (Eiríksson, 1980). These rocks were termed the Tjörnes beds before Eiríksson et al. (2021b) formally defined them as the Barmur Group. The overlying Höskuldsvík Group comprises mainly basaltic lavas intercalated with fluvial sedimentary rocks, which record a sub-aerial environment. At the top of the sequence, the Breiðavík Group consists of basaltic lavas intercalated with sedimentary rocks including diamictites, which show that the Breiðavík Group was deposited during Pleistocene glacial conditions (Eiríksson, 1985; Eiríksson & Geirsdóttir, 1991; Eiríksson et al., 1990, 1992, Eiríksson, Guðmundsson, et al., 2021; Eiríksson, Knudsen, & Símonarson, 2021; Geirsdóttir & Eiríksson, 1994). The entire Tjörnes sedimentary succession is deposited in an extensional basin formed within the inside corner between the Tjörnes Fracture Zone and the Northern Volcanic Zone rift axis (Eiríksson, 1980, 1981, 1985). Tectonic uplift during the Late Pleistocene produced the coastal exposure observed today, though deposition continues in the adjacent Skjálfandi basin (Einarsson et al., 1967; Sæmundsson, 1974; Thoroddsen, 1902).

Kaldakvísl Group lavas were dated as 9.9 ± 1.8 and 8.6 ± 0.4 Ma (late Miocene) by Aronson and Sæmundsson (1975) using the K-Ar method. The contact between the Kaldakvísl Group basement and the Barmur Group sedimentary succession is an erosional unconformity that is cut by a strike-slip fault zone where it is exposed in the Kaldakvísl river valley (Figure 1 of Einarsson et al., 1967; Eiríksson, 1980). To the north of the main contact, two basaltic lava flows were originally interpreted to be intercalated within the basal sediments of the Barmur Group (Bárðarson, 1925; Eiríksson, 1980, 1981). The uppermost of these lavas has been dated as 4.3 ± 0.4 Ma using the K-Ar method (Albertsson, 1976). More recent mapping indicates that these lower lavas may instead be faulted slivers of the Kaldakvísl Group (Eiríksson et al., 2021b).

The pioneering stratigraphic study of the Tjörnes beds, as the Barmur Group was then called, was conducted by Bárðarson (1925), who divided them into 25 shelly layers (labeled 1–25) and 10 terrestrial layers (labeled A–J). Bárðarson (1925) further assigned the beds into three bivalve molluscan biozones: the Tapes Zone (beds 1-5), the Mactra Zone (beds 6–12), and the Serripes Zone (beds 13–25). The boundary between the Mactra/Serripes biozones is of key importance to our study because it represents the sudden arrival of abundant Pacific-type fauna in Iceland. A few Pacific-type molluscs appear in the Barmur Group below the Serripes biozone (Símonarson & Eiríksson, 2021b). The abrupt appearance of at least 24 new Pacific-type mollusc species in shell bed 13 that were not present in bed 12 or lower defines the base of the Serripes biozone, and this bioevent is the widely used signal of the first opening of the Bering Strait oceanic gateway (Buchardt & Simonarson, 2003; Símonarson & Eiríksson, 2008, 2021a; Verhoeven et al., 2011). Importantly, the shallow marine sediments c. 17 m above the Mactra/Serripes biozone boundary contain intercalated basaltic pillow lavas surrounded by palagonitic tuff and breccia, exposed inland in the Skeifá river valley (Eiríksson et al., 2021b). Albertsson (1976) attempted twice to measure the absolute K-Ar age of the Skeifá basalts; one attempt was inconclusive, and a second yielded an age of 2.46 ± 0.46 Ma that was considered erroneously young. Nearly a century after the pioneering work, and after four decades of research, Eiríksson et al. (2021b) presented a comprehensive lithostratigraphic scheme for the Barmur Group that addresses gaps in the stratigraphy and correlates the coastal and inland exposures (Figures 1c and 2). Retaining the molluscan biozonation of Bárðarson (1925), they divide the Barmur Group into 10 formations, 23 members and 219 lithological units, with an aggregate thickness exceeding 575 m (Figure 1c).

Most studies consider that sediment and lava accumulation was fairly continuous across the boundary between the Barmur and Höskuldsvík Groups, although Verhoeven et al. (2011) propose an erosional hiatus of c. 0.6 My between the groups (Figure 2). Basaltic lavas of the Höskuldsvík Group have been dated as 2.55 ± 0.27 , 2.36 ± 1.8 , and 3.19 ± 0.26 Ma using the K-Ar method (Albertsson, 1976, 1978). Eiríksson et al. (2021a) considered the younger ages unreliable and based their age model for the group on the c. 3 Ma age (late Pliocene). There is an angular unconformity between the Höskuldsvík Group and overlying Breiðavík Group (Eiríksson, Guðmundsson, et al., 2021). The age of the Breiðavík Group is well-established to be Quaternary, post iNHG, based on the sedimentary record of glacial cycles covering the Icelandic lowlands, and Ar-Ar age determinations in the upper part of the succession correlated with a large amount of good quality paleomagnetic data (Camps et al., 2011; Eiríksson et al., 1990, 2021a).

3. Materials and Methods

3.1. Fieldwork

Sampling of four basaltic lavas was undertaken during September 2018 and March 2019 (Figures 1b and 1c). In the Skeifá river valley, we sampled the lowermost of the two Skeifá pillow lavas (Unit XIV:P of Eiríksson et al., 2021b) in order to date the appearance of abundant Pacific-type bivalve molluscs which lies c. 17 m beneath,

and by inference the opening of the Bering Strait. At Höskuldsvík, we targeted the oldest of the six lavas in the Höskuldsvík Group (Unit XXII:J of Eiríksson et al., 2021b) to constrain the age of the top of the Barmur Group and determine whether it records the mPWP. We sampled two separate lava flows either side of the Kaldakvísl river mouth to constrain the age of the base of the Barmur Group. To the south of the river mouth, we targeted the uppermost lava on Lynghöfði hill (hereafter "Lynghöfði lava"), considered one of the youngest exposed flows in the Kaldakvísl Group basement to the Barmur Group (Eiríksson et al., 2021b). To the north of the river mouth, we sampled the uppermost of the two lava flows exposed on Grænhöfði hill (hereafter "Grænhöfði lava"; Unit V:A of Eiríksson et al., 2021b), which was originally thought to lie within the Barmur Group (Bárðarson, 1925; Eiríksson, 1980, 1981), but was recently hypothesized to be a faulted sliver of the Kaldakvísl Group basement (Eiríksson et al., 2021b). All these lavas were excavated using a 1.8 kg long-handled lump hammer and cold chisels to remove weathered surfaces and to expose minimally altered basalt at the centre of each flow. Fresh and dense hand samples were collected at each field locality. Loose surficial material was removed using a scrubbing brush, before the samples were packaged in labeled cloth bags.

3.2. ⁴⁰Ar/³⁹Ar Dating

The basalt samples were initially thin sectioned at the University of Birmingham, UK and screened using a petrographic microscope to identify a subset with minimal signs of alteration in the groundmass (see Supporting Information S1). Selected hand samples were then prepared for 40 Ar/ 39 Ar dating at the Scottish Universities Environment Research Centre (SUERC, East Kilbride; UK). Visibly altered areas were removed using a lapidary trim saw. The samples were subsequently crushed using a club hammer to yield 2 mm fragments, before being ground in a disc-mill and sieved to produce a 250–500 µm separate. Following a rinse in deionized water to remove the fine fraction, the samples were leached in an ultrasonic bath at 50°C for 1 hr, first in 3N HNO₃ and then in 3N HCL to remove any trace alteration phases. This dual-acid leaching process was repeated at least three times until the acids remained clear. The leached samples were then ultrasonicated in deionized water, gently dried (temperature <50°C) and passed through a Frantz isodynamic magnetic separator to remove phenocrysts, xenocrysts and any remaining alteration products from the groundmass. The samples were then handpicked under a binocular microscope to harvest a homogenous and purified groundmass separate. Each analyzed sample represented approximately 120 mg of groundmass separate.

Samples and neutron flux monitors were packaged in Cu foil and stacked in quartz vials with the relative positions of packets precisely measured for later reconstruction of neutron flux gradients. The sample package was irradiated in the Oregon State TRIGA Reactor (Corvallis, Oregon; USA), Cd-shielded facility (CLICIT). Alder Creek sanidine with an established age of 1.1891 ± 0.0008 (1 σ) Ma (Niespolo et al., 2017) was used to monitor ³⁹Ar production and to establish neutron flux values (J) for the samples. At SUERC, gas was extracted from the samples via step-heating using a mid-infrared (10.6 μ m) CO₂ laser with a non-Gaussian, uniform energy profile and a 3.5 mm beam diameter rastered over a sample well. The samples were housed in a doubly pumped ZnS-window laser cell and loaded into a copper planchette containing four square wells (each measuring 1.6 cm²). Liberated argon was purified of active gases (e.g., CO₂, H₂O, H₂, N₂, CH₄) using three Zr-Al getters: one at 16°C and two at 400°C. Data for the four groundmass samples were collected on a GVI ARGUS V multi-collector mass spectrometer using a variable sensitivity faraday collector array in static collection (non-peak hopping) mode (Mark et al., 2009). Time-intensity data were regressed to inlet time with second-order polynomial fits to the data. The average total system blank for laser extractions, measured between each sample run, was $6.4 \pm 0.1 \times 10^{-15}$ mol ⁴⁰Ar, $11.0 \pm 0.8 \times 10^{-17}$ mol ³⁹Ar, and $2.4 \pm 0.2 \times 10^{-17}$ mol ³⁶Ar. Mass discrimination was monitored on a daily basis, between and within sample runs by analysis of an air standard aliquot delivered by an automated pipette system (D values applied to individual steps are listed in the data set hosted on Zenodo). All blank, interference and mass discrimination corrections and age calculations were performed with MassSpec software (MassSpec, v.8.058, authored by Al Deino, Berkeley Geochronology Center). Decay constants and corrections are after Renne et al. (2010).

Each sample was run in duplicate or triplicate, with each single analysis converted into a plateau age (Figure 3). Plateau ages were accepted when at least three contiguous age steps are indistinguishable at the 2σ level, contain a minimum of 50% of the total ³⁹Ar released during the experiment, and define an inverse isochron indistinguishable from the plateau age at 2σ uncertainty. Additionally, the trapped component composition, derived from the inverse isochron, should be indistinguishable from air at 2σ . Age and uncertainty were calculated using the mean







weighted by the inverse variance of each step. The accepted plateau steps from the duplicate or triplicate runs were combined, and a weighted mean was calculated yielding a final "composite age," which is accepted as the best age estimate for the sample. Composite ages are presented at 2σ including uncertainty in J; uncertainties with and without J are presented in the data set hosted on Zenodo.

3.3. Magnetostratigraphy

We correlated our new ⁴⁰Ar/³⁹Ar ages with existing paleomagnetic data to refine the age model for the Barmur Group. We use the Geomagnetic Polarity Time Scale of Ogg (2020) throughout. A comprehensive review of the available paleomagnetic data from the entire Tjörnes succession is provided by Eiríksson and Símonarson (2021) and Eiríksson et al. (2021a). Relevant paleomagnetic data were presented by Hospers (1953), Doell (1972), Gladenkov and Gurari (1976), Eiríksson et al. (1990), and Kristjánsson (2004). The data are not of uniform stratigraphic distribution, nor of uniform quality. Kristjánsson (2004) reported 31 generally good quality measurements from the upper part of the Barmur Group, which together with the data set of Doell (1972) have allowed us to confidently correlate our radiometrically dated samples from the Höskuldsvík and Skeifá lavas with unique magnetic subchrons. Minimal paleomagnetic data is available for the lower part of the Barmur Group.

4. Results

4.1. Age of the Skeifá Lava and the Pacific-To-Atlantic Mollusc Migration

The lowermost flow in the Skeifá pillow lavas, intercalated within the Barmur Group, has an 40 Ar/ 39 Ar composite plateau age of 5.06 ± 0.16 Ma, in agreement with an isochron age of 5.12 ± 0.19 Ma (Figure 3). The Skeifá lavas are considered reversely magnetized (Eiríksson et al., 2021a). Doell (1972) obtained a reliable reverse polarity measurement from the upper Skeifá lava (denoted Unit I37). Kristjánsson (2004) reported four out of six samples of the lower Skeifá lava (denoted SA1) were reversed, though uncertainties on all six readings were among the largest in his data set. However, all available data sets give reliable reverse polarity readings for the sedimentary rocks immediately surrounding the Skeifá lavas (Doell, 1972; Gladenkov & Gurari, 1976; Kristjánsson, 2004), and there is no field evidence for significant breaks between sedimentary deposition and lava emplacement (Eiríksson et al., 2021b). Thus we accept the reverse designation of our radiometrically dated Skeifá lava. The age range we determined is 5.31–4.90 Ma (2 σ range for both the plateau and isochron ages). The only reverse polarity interval to overlap the radiometrically determined range is subchron C3n.3r, at 4.997–4.896 Ma (Figure 4). We therefore assign the lower Skeifá lava, and by inference the entire Skeifá lava package, to magnetic subchron C3n.3r. These lavas were emplaced during the Zanclean stage (Early Pliocene), significantly earlier than has been suggested in all previous age models (Figure 2).

The *Mactra/Serripes* bivalve mollusc biozone boundary marks the abrupt appearance of abundant Pacific-type fauna within the Barmur Group (Durham & MacNeil, 1967; Einarsson et al., 1967; Eiríksson et al., 1990; Símonarson & Eiríksson, 2008, 2021a). This bioevent is the signature of Bering Strait gateway opening. The *Mactra/Serripes* boundary is observed c. 17 m below the Skeifá pillow lavas at Tungugrænur (Eiríksson et al., 2021b). Eiríksson et al. (2021b) placed the *Mactra/Serripes* boundary within the same reverse polarity chron as the Skeifá lavas. Considering the distribution of mollusc fossils in the Tungugrænur section, the *Mactra/Serripes* boundary could also lie within the underlying normal polarity chron. Thus, we assign the *Mactra/Serripes* boundary to magnetic subchrons C3n.4n–C3n.3r (5.235–4.896 Ma). In detail: reversely magnetized paleomagnetic sample set SA-1 (Kristjánsson, 2004) is from Skeifá lavas designated Units XIV:P–S in the lithostratigraphic scheme of Eiríksson et al. (2021b); reversely magnetized sedimentary samples SA-2, SA-3, and SA-4 lie within Units XIV:O, XIV:N–J, and XIV:I respectively; and normally magnetized sedimentary sample SA-5 also lies within Unit XIV:I (J. Eiríksson, *pers. comm.*). Eiríksson et al. (2021b) placed the *Mactra/Serripes* biozone boundary at the boundary between Units XIV:I and XIV:I, but the biozone boundary could lie anywhere between mollusc fossil occurrences within basal Unit XIV:I and Unit XIV:L (Figure 5.54 of Eiríksson

Figure 3. ⁴⁰Ar/³⁹Ar age spectra and inverse isochrons for the Lynghöfði, Grænhöfði, Skeifá, and Höskuldsvík lava samples. The left panel shows step-heating plateau age spectra with $\%^{40}$ Ar*, K/Ca and age versus $\%^{39}$ Ar released per step for each color-coded replicate. The ages presented in the plateau age spectra are calculated using air as a trapped component. Box heights are 2σ . The right panel shows isochron analyses which yield an age and a composition of the trapped component for a number of points (*n*). The isochrons include a black regression line, dashed 2σ error envelope and colored circles, which represent the points that make up the respective plateau ages for each replicate. The mean square weighted deviation and P (Chi-squared value) are also shown.





Figure 4. A new age model for the Barmur Group constructed using our 40 Ar/³⁹Ar ages and the reinterpretation of existing paleomagnetic records (Doell, 1972; Kristjánsson, 2004), correlated to the Geomagnetic Polarity Time Scale of Ogg (2020). Paleomagnetic sampling positions and numbering system of Doell (1972) and Kristjánsson (2004) are shown relative to their respective stratigraphic horizons after Bárðarson (1925) and Eiríksson et al. (2021b). The legacy K-Ar ages of Albertsson (1976, 1978) are also shown for reference. Composite lithostratigraphic column and molluscan biozonation as in Figure 1. 40 Ar/³⁹Ar composite age abbreviations as in Figure 2 and 3.

et al., 2021b; J. Eiríksson, *pers. comm.*) Thus, the biozone boundary could lie within the reversely magnetized section XIV:L–I, which is assigned to subchron C3n.3r by dating the Skeifá lavas, or it could lie within the underlying normally magnetized section XIV:I–H, which we assign to subchron C3n.4n (Figure 4). In summary, the bioevent which signifies Pacific-to-Atlantic faunal migration through the Bering Strait occurred between 5.2 and 4.9 Ma during the Zanclean stage (Early Pliocene), significantly earlier than has been suggested in all previous age models (Figure 2).

4.2. Age of the Upper Barmur Group

The lowermost lava in the Höskuldsvík Group, immediately overlying the Barmur Group, yields an ${}^{40}\text{Ar}/{}^{39}\text{Ar}$ composite plateau age of 4.43 ± 0.12 Ma, in good agreement with the calculated isochron age of 4.40 ± 0.19 Ma (Figure 3). This result is significantly older than the K-Ar age of 2.55 ± 0.27 Ma reported for the same lava (Albertsson, 1976). Given the well-known susceptibility of the K-Ar technique to argon loss, questions over the quality of the Höskuldsvík Group samples raised by Albertsson (1976), and that Albertsson (1976) also reported a relatively older K-Ar age of 3.19 ± 0.26 Ma for stratigraphically younger lavas within the Höskuldsvík Group, we are confident that the new ${}^{40}\text{Ar}/{}^{39}\text{Ar}$ age of c. 4.4 Ma is more reliable than the previous age determinations.

Normal polarity magnetism was measured by Doell (1972) in eight samples collected from the flow that we dated (denoted I38). Our radiometrically determined age range is 4.59-4.21 Ma (2σ range for both the plateau and isochron ages). This range overlaps with both the C3n.2n (Nunivak) subchron at 4.631-4.493 Ma and the C3n.1n (Cochiti) subchron at 4.300-4.187 Ma. We prefer to assign the lowermost Höskuldsvík Group lava to the C3n.2n subchron (4.631-4.493 Ma), which overlaps with the uncertainty ranges of both the composite plateau and isochron ages, whereas the alternative subchron would overlap with the isochron age range only. This choice is supported a posteriori by the correlations below.

Having correlated the Skeifá lava with subchron C3n.3r and the lowermost Höskuldsvík Group lava with subchron C3n.2n (4.631-4.493 Ma), we now correlate the Barmur Group sediments between them with the paleomagnetic timescale (Figure 4). The lowermost Höskuldsvík lava conformably overlies the Barmur Group. There is no evidence of erosion (Eiríksson et al., 2021b). Furthermore, the basal part of the lava flow is brecciated and contains inclusions of underlying sediment, implying that the Barmur Group substrate was unlithified at the time of lava emplacement (Eiríksson et al., 2021b), and suggesting that the uppermost sediments of the Barmur Group are of similar age to the lava. The quality of paleomagnetic data for the uppermost Barmur Group bed (above bed J) is relatively poor, but these beds are interpreted as normally magnetized (Kristjánsson, 2004). We therefore assign them to the C3n.2n subchron, together with the Höskuldsvík Group lava we dated. The underlying Barmur Group sediments within the Serripes biozone that are exposed along the coast are a conformable package (Eiríksson et al., 2021b). Kristjánsson (2004) reported reversed polarity between Barmur Group beds I and J, near the top of the succession, which we correlate with subchron C3n.2r (4.799–4.631 Ma). Beneath this, he reported an interval of normal polarity in the coastal section between Barmur Group beds 13 and H, which we correlate with subchron C3n.3n (4.896-4.799 Ma). The coastal exposure described above is not continuous with the inland exposure which includes the Skeifá lavas. Eiríksson et al. (2021b) carried out a detailed program of logging and mapping to correlate the two exposures. They place the Skeifá pillow lavas and surrounding sediments between beds 13 and 12 in their composite lithostratigraphic column (Figure 1c). The inland Skeifá river valley section correlates with an erosional unconformity beneath bed 13 at the coast, which explains why the lavas are not exposed there. This scheme places the reversely magnetized Tungugrænur succession that includes the Skeifá lava and the *Mactra/Serripes* boundary directly below the coastal section we have correlated with subchron C3n.3n (Figure 4). Thus, the Skeifá lavas lie within subchron C3n.3r and the Mactra/Serripes boundary within subchron C3n.3r or C3n.4n, as we showed independently in Section 4.1. In summary, the Serripes biozone of the Barmur Group contains a record of each and every magnetic subchron between the Skeifá and Höskuldsvík lavas we have dated.

4.3. Age of the Lower Barmur Group

The Lynghöfði lava, thought to be one of the youngest lavas exposed in the Kaldakvísl Group basement to the Barmur Group (Eiríksson et al., 2021b), yields an 40 Ar/ 39 Ar composite plateau age of 9.754 ± 0.11 Ma in agreement with an isochron age of 9.78 ± 0.13 Ma (Figure 3). Our new age is reasonably well in line with

25724525, 2023, 4, Down

1029/2022PA004539 by

the K-Ar ages from the Kaldakvísl Group obtained by Aronson and Sæmundsson (1975), which ranged from 9.9 ± 1.8 to 8.6 ± 0.4 Ma, though the 40 Ar/ 39 Ar age is more reliable. Thus we confirm previous estimates that the Kaldakvísl Group is Tortonian (Late Miocene) in age.

The Grænhöfði lava, which may lie below or within the Barmur Group, yields an 40 Ar/ 39 Ar composite plateau age of 9.29 ± 0.087 Ma in agreement with the isochron age of 9.27 ± 0.11 Ma (Figure 3). Hospers (1953) detected normal polarity in this flow, so we correlate it with subchron C4Ar.1n (9.426–9.311 Ma). Albertsson (1976) reported a K-Ar age of 4.30 ± 0.17 Ma for the same lava but he considered this to be a minimum (youngest possible) age. Our more reliable 40 Ar/ 39 Ar age justifies Albertsson's caution. We find that the Grænhöfði lava is about 0.5 My younger than the Lynghöfði lava. Thus, our results support the hypothesis of Eiríksson et al. (2021b) that the Grænhöfði lava is not part of the Barmur Group, but instead is a faulted sliver of the Kaldakvísl Group basement.

The contact between the Kaldakvísl Group and the overlying Barmur Group is an erosional unconformity (Eiríksson et al., 2021b). Weathering and reddening of the Kaldakvísl Group beneath the contact implies a significant hiatus in deposition. Our new ages show that the maximum (oldest possible) age for the base of the Barmur Group is c. 9.3 Ma, but it may well be significantly younger. There are no other lavas that are undoubtedly part of the Barmur Group beneath the lowest Skeifá lava, which we have dated as 5.06 Ma. Thus, the only way to use radiometric dating of lavas to estimate the age of the base of the Barmur Group is to estimate the sedimentation rate in the upper Barmur Group and extrapolate it downwards. Eiríksson et al. (2021b) demonstrate that 217 m of sediments occur between the lowermost Skeifá lava and the basal lava of the Höskuldsvík Group. Our ⁴⁰Ar/³⁹Ar data show the time interval between these two lavas is 0.63 ± 0.28 Ma, which equates to a sedimentation rate of 0.03 cm/yr (0.02–0.06 cm/year accounting for analytical error). We follow previous studies in assuming a fairly consistent rate of sedimentation for the entire Barmur Group (Albertsson, 1978; Buchardt & Simonarson, 2003; Einarsson et al., 1967; Símonarson & Eiríksson, 2008). The stratigraphic thickness between the lowermost Skeifá lava and the base of the Barmur Group resting on the erosional surface at Grænhöfði is 308 m (Eiríksson et al., 2021b). Linear extrapolation then gives an age of 5.95 ± 0.56 Ma (terminal Miocene) for the base of the Barmur Group (Figure 4). The corresponding hiatus represented by the unconformity at the base of the Barmur Group is 3.34 ± 0.64 My. Our extrapolated terminal Miocene age for the basal Barmur Group agrees with the estimated age of 5.85 ± 0.5 Ma proposed by Eiríksson et al. (2021a), and is compatible with the occurrence of late Miocene foraminiferal species, such as *Cibicidoides limbatosuturalis*, in the lower section of the *Tapes* biozone (Knudsen et al., 2021). Our age estimate for the earliest sediments is also broadly in line with the basin formation mechanism. The Tjörnes Basin exists within the inside corner of the Northern Volcanic Zone plate spreading axis and the Tjörnes Fracture Zone, which had formed by c. 7 Ma (Garcia et al., 2003; Sæmundsson, 1974; Sigmundsson et al., 2020).

4.4. Age of the Höskuldsvík Group

Verhoeven et al. (2011) proposed that a hiatus of 0.6–1.0 My exists between the Barmur Group and the overlying Höskuldsvík Group, in contrast to all other age models (Figure 2). Our radiometric dating shows a time interval of about 0.6 My between the Skeifá lava within the Barmur Group and the basal lava of the Höskuldsvík Group, and the sedimentary rocks between these tie points record each and every magnetic subchron. We therefore doubt that a significant hiatus occurs at the top of the Barmur Group. Our age for the base of the Höskuldsvík Group is up to 1 My older than previous estimates (Figure 2), implying the overlying Tjörnes succession accumulated over a considerably longer period of time than has previously been proposed. The implications for the Icelandic record of lowland glaciation are discussed in Section 5.4.

5. Discussion

5.1. Pacific-To-Atlantic Faunal Migration

Mollusc assemblages in the Pacific and Atlantic/Arctic realms were distinct between the middle Cretaceous and the middle to late Miocene, indicating that the Bering Strait oceanic gateway did not yet exist, and a land-bridge connected Asia with North America (Marincovich & Gladenkov, 2001). Opening of the Bering Strait gateway is signaled by the first appearance of Atlantic bivalve molluscs in the Pacific, and vice versa. We have dated the arrival of abundant Pacific bivalves in Iceland in the North Atlantic to 5.2–4.9 Ma, during the early Pliocene

HALL ET AL.





Figure 5. Summary of late Neogene trans-Arctic molluscan migration models, showing a schematic comparison between: (a) the two-stage late Miocene to early Pliocene (5.6–4.8 Ma; dark mahogany arrows) and mid-Pliocene (3.6 Ma; dark teal arrows) event proposed by Marincovich (2000) and (b) the single-stage late Miocene to early Pliocene event (arriving in Iceland at c. 5.2–4.9 Ma) advocated herein. A single and earlier phase of trans-Arctic interchange is further supported by the distribution of late Miocene/early Pliocene Pacific diatoms in the Chukchi Sea and strontium dating of Pacific molluscs preserved at Heumensoord, Netherlands (Beets, 1992; Gladenkov & Gladenkov, 2004). Late Neogene continental outlines created in QGIS v.3.22 following the tectonic reconstructions of Torsvik et al. (2002). Light red, orange, and blue arrows indicate surface, intermediate, and deep ocean currents, respectively.

(strictly, to magnetic subchrons C3n.4n–C3n.3r, at 5.235–4.896 Ma). Our age for this faunal exchange event is over 1.3 million years earlier than previous estimates (Buchardt & Simonarson, 2003; Einarsson et al., 1967; Eiríksson et al., 2021a; Símonarson & Eiríksson, 2008). Most of the previous younger age estimates were underpinned by 1970s-vintage K-Ar radiometric ages. The K-Ar dating method is well known to underestimate the age of basalts that have low initial K contents or are subject to radiogenic Ar loss through hydrothermal alteration, as observed in the Tjörnes succession (Aronson & Sæmundsson, 1975). In such cases, it is within expectation that modern ⁴⁰Ar/³⁹Ar dating methods should return ages up to several million years older than respective vintage K-Ar dates (McDougall & Harrison, 1999).

Marincovich (2000) proposed an age of 3.6 Ma for the Pacific-to-Atlantic faunal exchange event by correlating the reported first appearance of the Pacific bivalve Mya arenaria at the Mactra/Serripes boundary in the Barmur Group, Iceland (Figure 5a), with its first appearances in the supposedly correlative Coralline Crag Formation of England and Rushmere Member (Yorktown Formation) of the eastern USA (Campbell, 1993; MacNeil, 1965). However, Símonarson and Eiríksson (2021a, 2021b) conducted a comprehensive study of the genus Mya and showed that the reports of Mya arenaria in the Barmur Group were mistaken; the specimens were actually Mya schwarzbachi. Furthermore, the Coralline Crag Formation has now been dated as 4.4–3.8 Ma (De Schepper et al., 2009) and the base of the Rushmere Member as 3.3 Ma (Dowsett et al., 2021), so first appearances of any fauna at these locations compared with the Barmur Group must be diachronous. Thus, the 3.6 Ma age estimate for the Pacific-to-Atlantic faunal migration should be discounted. Verhoeven et al. (2011) dated the sudden arrival of Pacific molluscs in the Barmur Group at around 4.5 Ma using dinoflagellate biostratigraphy (Figure 2). We consider their estimate is a minimum (youngest possible) age because the age-diagnostic Reticulatosphaera actinocoronata extends well above the Mactra/Serripes boundary and up to Barmur Group bed 17, and has a known highest occurrence of 5.05-4.44 Ma in the Nordic Seas (Channell et al., 1999; Poulsen et al., 1996). Available biostratigraphic dating constraints for the Barmur Group are therefore in line with our radiometric estimate of 5.2-4.9 Ma for the Pacific-to-Atlantic faunal exchange event. Other lines of biostratigraphic evidence

25724525, 2023, 4, Dow

from the Arctic and north Atlantic support a latest Miocene to earliest Pliocene age for Pacific-to-Atlantic biotic exchange (Figure 5b). Pacific-to-Arctic migration of floral elements through the Bering Strait during the time period 5.6–4.8 Ma has been inferred from the diatom distribution in a vibro-cored section from the Chukchi Sea, close to the north of the Bering Strait (Gladenkov & Gladenkov, 2004). Beets (1992) determined a ⁸⁷Sr/⁸⁶Sr age of 4.8–4.7 Ma for the first appearance of Pacific molluscs (including *Mya truncata*) in Pliocene sediments from the Heumensoord-46A/260 borehole in the Netherlands (Meijer, 1993).

The first opening of the Bering Strait has previously been dated by observations of Arctic-to-Pacific faunal exchange. Marincovich and Gladenkov (2001) estimated that the latest possible age for the first appearance of the Arctic bivalve *Astarte* in the Bear Lake Formation at Sandy Ridge, southwestern Alaska (i.e., in the northern Pacific), was 5.5–4.8 Ma. The dating was done using age-diagnostic diatoms excavated from within the bivalve shells. The age range is compatible with estimated ages of the first appearances of *Astarte* in other northern Pacific sections from Kamchatka and Japan (Gladenkov & Gladenkov, 2004). The published range of 5.5–4.8 Ma uses the Berggren et al. (1995) timescale; the equivalent age range within the Ogg (2020) timescale we use here is 5.6–4.8 Ma. Gladenkov et al. (2002) refined the most likely age of Arctic-to-Pacific mollusc faunal exchange to 5.5–5.4 Ma on the Berggren timescale (5.6–5.4 Ma on the Ogg timescale) using evidence from the Milky River Formation at Sandy Ridge. Thus, our new age range of 5.2–4.9 Ma for arrival of abundant Pacific mollusc fauna in Iceland is similar to, but does not quite overlap, the most reliable published age range of 5.6–5.4 Ma for Arctic-to-Pacific migration.

Marincovich (2000) proposed a two-stage faunal exchange model in which Arctic-to-Pacific migration occurred on first opening of the Bering Strait by the early Pliocene, followed several million years later by Pacific-to-Atlantic migration during the late Pliocene (Figure 5a). Our results cast doubt on this model (Figure 5b). If the age ranges are taken at face value, the arrival of abundant Pacific molluscs in Iceland in the Atlantic (5.2-4.9 Ma) could have occurred 0.2-0.7 million years after the arrival of Arctic molluscs at Sandy Ridge in the Pacific (5.6-5.4 Ma). However, the discussion above showed that the first recorded arrivals of Pacific-type mollusc fauna at different sites in the North Atlantic and Arctic vary by up to 1.9 million years (compare our bound of 5.2 Ma for the Barmur Group with the bound of 3.3 Ma for the Rushmere Member of the Yorktown Formation from Dowsett et al., 2021). Given this evidence for diachroneity, it is probably unwise to interpret trans-Arctic faunal exchange in two distinct stages on the basis of just two widely separated stratigraphic successions. The difference in age range between the faunal events in Iceland and Alaska could reflect a period of several hundred thousand years for molluscs to cross the Arctic, and it could also reflect site-specific limitations on the occurrence of molluscs. Presence of sparse Pacific-type mollusc fauna within the Tapes and Mactra biozones of the Barmur Group (Símonarson & Eiríksson, 2021b) is in line with the hypothesis that the molluscs crossed the Arctic quickly after first gateway opening, but the first appearance of abundant Pacific-type molluscs in Iceland was moderated by site-specific conditions. Thus, the most parsimonious explanation for the available faunal exchange data is that first opening of the Bering Strait in the latest Miocene (c. 5.5 Ma) was soon followed by marine faunal exchange in both directions.

Though the two-stage model of Marincovich (2000) was based primarily upon the previously accepted late Pliocene estimates for arrival of Pacific molluscs in Iceland, it was apparently supported by general circulation model (GCM) experiments that predict southward surface currents through the Bering Strait following its first opening (Maier-Reimer et al., 1990). Initial southward flow was hypothesized to have prevented Pacific-to-Atlantic faunal exchange, which began only after flow reversed to its present northward direction, possibly related to closure of the Central American Seaway (Horikawa et al., 2015; Marincovich, 2000). We do not necessarily reject the flow patterns predicted by such circulation models because we doubt they can be confidently related to planktonic faunal dispersal patterns: GCM palaeoclimate models predict currents averaged over time periods of over 500 years, very considerably longer than the planktonic larval stage of bivalve molluscs which lasts days to months.

5.2. Bering Strait Gateway After Its Initial Opening

The long-term tectonic driver for the first Cenozoic opening of the Bering Strait oceanic gateway is subduction of the Pacific Plate beneath Alaska (Marincovich et al., 1990; Scholl et al., 1992). At present, flat-slab subduction of the Yakutat block (an oceanic plateau within the northeastern Pacific Plate) is causing high crustal strain rates and high topography in southeastern Alaska (Elliott & Freymueller, 2020; Redfield et al., 2007). Present-day

lateral extrusion of Alaskan crust away from this high-strain region and toward the west and south (i.e., escape tectonics) is indicated by GPS measurements (Elliott & Freymueller, 2020). These crustal motions are associated with extensional earthquake focal mechanisms on normal faults in the Bering Sea region at present, including in the Bering Strait itself (Elliott & Freymueller, 2020). Regional tectonic models suggest that the northward motion of the Pacific Plate first brought the Yakutat block into contact with the Aleutian subduction zone during the late Miocene (Redfield et al., 2007). Thus, the establishment of the modern tectonic scenario in the Alaska-Bering Sea region in the late Miocene was soon followed by opening of the Bering Strait gateway near the Miocene/ Pliocene boundary (by c. 5.5 Ma). The plate motion scenario and tectonic configuration do not appear to have changed significantly since the late Miocene (Redfield et al., 2007). Thus, the early Pliocene to present.

Nevertheless, there is abundant, long-standing evidence that the Bering Strait gateway closed and re-opened multiple times after its first opening c. 5.5 Ma (e.g., Hopkins, 1967). These cycles of re-opening are thought to be eustatically driven. At present, the gateway is c. 85 km wide and up to 55 m deep (Woodgate & Aagaard, 2005). The rise in eustatic sea-level following the last glacial maximum was approximately 130 m (Lambeck, 2004). Thus, eustatic sea-level changes associated with the Pleistocene glacial cycles temporarily closed the gateway and re-established the Bering land-bridge. There is a considerable literature on migrations of land animals, including humans, across the Pleistocene land bridge (e.g., Bennett et al., 2021; Goebel et al., 2008; Hopkins, 1959; Moreno-Mayar et al., 2018). Climate modeling studies suggest that an open Bering Strait gateway tends to cool the North Atlantic-Arctic region and promote formation of sea-ice, in comparison with a closed gateway scenario (Brierley & Fedorov, 2016; Hu et al., 2010). This result raises the question of the relationship between the history of Bering Strait opening and Plio-Pleistocene climate cooling, including the iNHG (Brierley & Fedorov, 2016; De Schepper et al., 2015). Before discussing this question, it is helpful to consider whether the gateway was predominantly open or closed during the period between its late Miocene initial opening at c. 5.5 Ma and the late Pliocene iNHG.

The bivalve molluscs used to signal Bering Strait opening have long temporal ranges, which is why they themselves are not used to date the opening (Marincovich & Gladenkov, 1999). The Pacific and Atlantic/Arctic faunas were distinct up to the Miocene because the Bering land bridge had existed continuously since the Cretaceous (Fiorillo et al., 2008; Símonarson & Eiríksson, 2021a). Closure of the Bering Strait during a glacial cycle, or potentially for several million years, would not allow time for the faunas to become distinct again. Thus, the mollusc migration bioevent is a proxy for the first opening of the Bering Strait gateway, and it provides no useful information on subsequent temporary closures.

Plio-Pleistocene marine sediments that crop out along the coast of Alaska either side of the Bering Strait provide a record of relative sea-level which can be used to reconstruct gateway opening. Three Plio-Pleistocene depositional episodes are recognized, termed the Colvillian (oldest), Bigbendian and Fishcreekian transgressions (Carter, Brigham-Grette, & Hopkins, 1986; Carter, Brigham-Grette, Marincovich, et al., 1986). Fossil molluscs show that all these sediments post-date initial gateway opening (Brigham-Grette & Carter, 1992). The marine packages crop out above modern sea-level, which is interpreted to indicate an open Bering Strait gateway during all three depositional phases (Kaufman & Brigham-Grette, 1993). The Colvillian sediments indicate temperatures warmer than the present, with no permafrost or sea-ice; temperatures were similar or slightly cooler during the Bigbendian; and air temperatures were significantly cooler during the Fishcreekian, though marine conditions remained warmer than present (Brigham-Grette & Carter, 1992). Unfortunately, the Alaskan Plio-Pleistocene sediments have limited potential for dating, and the mollusc migration bioevent itself has been used as a tie point. Until now, this bioevent has been assumed to occur at c. 3.0-3.5 Ma (Gladenkov, 1981; Hopkins, 1967, 1972). Since the Colvillian transgression occurred after gateway opening, its age was interpreted as 2.7–2.48 Ma, and the age of the Fishcreekian as 2.48-2.14 Ma (Brigham-Grette & Carter, 1992). When our new estimate of c. 5 Ma for the mollusc migration tie point is applied instead, the Colvillian transgression could easily have occurred before 3 Ma, and possibly during the early Pliocene. The evidence for relatively warm surface sea-water temperatures during the Colvillian and Bigbendian periods is in line with age assignments between 5 and 3 Ma, because the Continental Offshore Stratigraphic Test boreholes in the Bering Sea record relatively warm temperatures during this period (Lagoe, 1992). In the Chukchi Sea, just north of the Bering Strait, the first sediments that overlie the Upper Cretaceous basement, and are thought to post-date gateway opening, are a marine succession deposited between 5 and 2 Ma (dated independently of molluscs) in a climate warmer than today (Gusev et al., 2009). In summary, the evidence from sedimentary successions close to the Bering Strait gateway suggests that the gateway was open for a significant part of the 5 to 2 Ma period following its initial opening, and this early open period preceded local climatic cooling. An unconformity overlies the Pliocene succession, and the overlying sediments are thought to be Middle to Upper Pleistocene in age (Brigham-Grette & Carter, 1992; Gusev et al., 2009). This evidence perhaps suggests that the first regression to close the Bering Strait gateway for a significant period after its initial opening occurred during the latest Pliocene to early Pleistocene.

A continuous history of Bering Strait gateway opening can be estimated from the eustatic sea-level curves. O'Regan et al. (2010, Figure 4f) presented a sea-level curve from oxygen isotopes corrected for global cooling (Lisiecki & Raymo, 2005; Miller et al., 2005; Zachos et al., 2001), and superimposed the level that would imply closure of the Bering Strait gateway (50 m below modern sea-level). Whilst this prediction of gateway opening history ignores any possible tectonic vertical motions or changes in dynamic support, it does align with the predictions based on the sedimentary record close to the gateway discussed above. The eustatic sea-level curve suggests that the gateway was predominantly open for almost all of the Pliocene and the early Pleistocene (before 5 to 2 Ma). More frequent and longer lasting closures during glacial periods began at c. 1.6 Ma, and the gateway was predominantly closed after 1 Ma.

5.3. Climatic Consequences of Bering Strait Opening

The extent to which global climate change is influenced by inter-ocean flow has received attention for decades (Haug & Tiedemann, 1998; Maier-Reimer et al., 1990). Brierley and Fedorov (2016) show a GCM modeling study of the effect of opening and closing the Bering Strait while keeping other parameters constant. An open Bering Strait allows relatively fresh Pacific water into the Arctic and North Atlantic, which reduces the strength of AMOC, and reduces high-latitude Atlantic surface temperatures. Hu et al. (2010) found an essentially similar result for glacial-interglacial cycles. Eustatic sea-level fall during the glacial periods temporarily closes the Bering Strait, increases the strength of AMOC, causes high-latitude Atlantic warming, and thus promotes northern hemisphere icesheet melt-back. On the bases of these climate modeling results and late Pliocene (post 3.2 Ma) estimates for Bering Strait gateway opening, Brierley and Fedorov (2016) hypothesized that opening of the Bering Strait could have been a significant influence on the iNHG. Our results argue against this hypothesis. The Bering Strait gateway was open by the latest Miocene c. 5.5 Ma and remained predominantly open during the Pliocene and into the early Pleistocene (Section 5.2). The iNHG involved significant expansion of northern hemisphere ice sheets in the late Pliocene and culminated in the establishment of glacial-interglacial cycles by the early Pleistocene. Local to Iceland, the iNHG is expressed as expansion of the Greenland icesheet from the southern uplands across northern Greenland c. 3.5-3.0 Ma (Maslin et al., 1998), and establishment of cycles of lowland glaciation in Iceland by c. 2.7 Ma (Eiríksson, Guðmundsson, et al., 2021; Geirsdóttir & Eiríksson, 1994). Thus, the Bering Strait gateway was open for over 2 million years prior to the iNHG. The AMOC, as indicated by the %Northern Component Water proxy and sediment drift accumulation rates south of Iceland, was relatively strong during the period between 5 and 2.7 Ma when the Bering Strait gateway was open (Parnell-Turner et al., 2015; Poore et al., 2006). The combined evidence argues against a simple, direct link between first opening of the Bering Strait gateway, northern hemisphere cooling and AMOC weakening.

At present, flow of Pacific water northward through the Bering Strait supplies about one third of the total fresh-water input to the Arctic (Woodgate & Aagaard, 2005). Introduction of relatively fresh water to the Arctic during periods with an open Bering Strait gateway is thought to promote the formation of sea ice. Both sea ice and relatively fresh sea water are exported from the Arctic to the Atlantic via the Fram Strait and the East Greenland Current (EGC). It has been proposed that formation of the EGC at 4.5 Ma, and coeval development of sea ice in the Iceland Sea, were driven by the initial opening of the Bering Strait (Clotten et al., 2019; De Schepper et al., 2015). This hypothesis was made on the basis of the Verhoeven et al. (2011) age model for the Barmur Group in Iceland, which places the Pacific-to-Atlantic mollusc migration bioevent at 4.5 Ma (De Schepper et al., 2015). We have shown that the mollusc migration bioevent occurred between 5.2 and 4.9 Ma, which preceded formation of the EGC and Iceland Sea sea-ice by at least 400 thousand years. Given the fragmentary record of sea-level change near the Bering Strait (Section 5.2), it could perhaps be argued that the gateway was too shallow for significant low salinity Pacific water to enter the Arctic Ocean when it first opened, even though mollusc larvae could migrate, and that input of Pacific water sufficient to form sea ice and initiate the EGC did not occur until 4.5 Ma. On the other hand, development of sea ice within the Arctic Ocean and Nordic Seas was diachronous. Seasonal sea ice was already present at the ACEX site in the central Arctic (Lomonosov Ridge) during the Upper Miocene

(O'Regan et al., 2010; Stein et al., 2016), before first opening of the Bering Strait gateway. When the Bering Strait gateway first opened at c. 5.5 Ma, no noticeable change in ice-rafted debris occurred at the ACEX site (O'Regan et al., 2010). Although seasonal sea ice appeared in the Iceland Sea at 4.5 Ma, seasonal sea ice did not occur along the Eurasian margins of the Arctic Ocean until c. 4.0 Ma (Knies et al., 2014). Thus, we see no compelling evidence for a simple, direct link between the first opening of the Bering Strait, development of sea-ice in the Arctic Ocean and Nordic Seas, and initiation of the EGC.

5.4. The Late Pliocene-Quaternary Record From the Tjörnes Succession

We have shown that the top of the Barmur Group is up to 1.5 million years older than previous estimates (Figure 2). This revision has significant implications for the interpretation of palaeoenvironmental data from the Barmur Group. The Barmur Group contains a variety of macrofossils and microfossils that have been used extensively for palaeoenvironmental reconstruction (Bárðarson, 1925; Buchardt & Simonarson, 2003; Cronin, 1991; Dowsett & Poore, 1991; Field et al., 2017; Pflug, 1959; Schwarzbach & Pflug, 1957; Símonarson et al., 2021; Strauch, 1972b; Verhoeven & Louwye, 2013; Verhoeven et al., 2013). In particular, sea-surface temperature estimates from this locality (previously known as the Tjörnes beds) were included in the first four generations (0–3) of the PRISM reconstruction of the mPWP (Cronin, 1991; Dowsett & Poore, 1991; Dowsett et al., 1994, 1996; Dowsett et al., 1999, 2010). We have dated the top of the Barmur Group to 4.631–4.493 Ma (Early Pliocene). The Barmur Group therefore does not contain the mPWP. Our result supports the removal of the Barmur Group/Tjörnes beds data from the latest PRISM4 sea surface temperature reconstruction (Dowsett et al., 2016; Robinson et al., 2018). However, the PRISM4 global topographic reconstruction shows a closed Bering Strait (Haywood et al., 2016, 2020). A more realistic topographic boundary condition for the mPWP would show an open Bering Strait (Section 5.2).

Our revised age for the Barmur Group also has a significant impact on the age model for the overlying Tjörnes succession, which is important because these rocks provide the only onshore exposure record of the onset of lowland glaciation in Iceland (Geirsdóttir & Eiríksson, 1994). The first evidence of lowland glaciation defines the boundary between the Höskuldsvík Group (cycles of basaltic lavas intercalated with sub-aerial, non-glacial sediments) and the Breiðavík Group (cycles of lavas intercalated with glacial diamictites). Our revised age model for the Barmur Group could be accommodated in several ways, which are not mutually exclusive. One possibility is that the Höskuldsvík Group accumulated continuously over a considerably longer period of time than has previously been proposed (Figure 2, Option 1). In this scenario the Höskuldsvík Group might record the mPWP. But although such an mPWP record might mitigate its absence in the Barmur Group in principle, the chances of obtaining useful palaeoenvironmental data from the Höskuldsvík Group sub-aerial lava field environment seem small. Another possibility is that a break in deposition occurred between the Höskuldsvík and Breiðavík Groups (Figure 2, Option 2). This possibility is supported by the observation of an angular unconformity between these groups (Eiríksson, Guðmundsson, et al., 2021). The related erosion might be linked to the first establishment of lowland glaciation in Iceland, since a 6 m thick lodgement till (Unit I:A of Eiríksson, Guðmundsson et al., 2021) rests directly upon the unconformity. A final possibility for the upper Tjörnes sequence age model is that the base of the Breiðavík Group could have a different age than currently thought (Figure 2, Option 3). The commonly accepted age is c. 2.6 Ma (Eiríksson et al., 2021a). This age is based on interpolation between Ar-Ar ages of <1.2 Ma from the upper Breiðavík Group (Camps et al., 2011) and the 1970s K-Ar ages for the Höskuldsvík Group that we have revised here. A similar interpolation with our Ar-Ar age for the Höskuldsvík Group would suggest the lowermost Breiðavík Group is up to 1 million years older than 2.6 Ma, potentially bringing the age of the Icelandic lowland glaciation into line with the northward expansion of ice in Greenland (3.5-3.0 Ma, Maslin et al., 1998), but this interpolation is not valid if an unconformity is present beneath the Breiðavík Group. Verhoeven et al. (2011) suggested an age of >1.95 Ma for the Hörgi Formation near the base of the Breiðavík Group on the basis of dinoflagellate cyst biostratigraphy, but they did not date the basal Breiðavík Group. Thus, the age of the onset of lowland glaciation in Iceland is not well constrained. We are currently in the process of testing these possibilities for the upper Tjörnes sequence age model by Ar-Ar dating of the lavas from the upper Höskuldsvík and lower Breiðavík Groups.

6. Conclusions

We obtained four new Ar-Ar radiometric ages for basaltic lavas that lie within, above and below the Barmur Group (formerly Tjörnes beds) of northern Iceland, and integrated them with existing paleomagnetic data, in order to better exploit the many existing records of palaeo-environmental change from this section.

25724525, 2023, 4, Downloaded

from https://agupub

.com/doi/10.1029/2022PA004539 by University Of Birmingham

And Serials Team, Wiley Online Library on [03/08/2023]. See the 7

Wiley Online Library for rules of

Our Ar-Ar age for the Skeifá lava shows that the first arrival of abundant Pacific bivalve molluscs in Iceland, which is widely accepted as a signal of first opening of the Bering Strait, occurred within magnetic subchrons C3n.4n–C3n.3r, at 5.235–4.896 Ma. This age estimate is over 1.3 million years earlier than previous estimates, which were all skewed by less reliable 1970s-vintage K-Ar ages. Our new age range for the first Pacific-to-Atlantic faunal exchange is similar to, but does not quite overlap, the most reliable published age range of 5.6–5.4 Ma for Arctic-to-Pacific migration. The difference in age might reflect site-specific limitations on mollusc habitability, since sparse Pacific molluscs have been noted prior to the arrival of abundant Pacific molluscs in Iceland, and it might also reflect the time taken for Pacific molluscs to migrate across the Arctic. Thus, first opening of the Bering Strait gateway in the latest Miocene, c. 5.5 Ma, was soon followed (within a few hundred thousand years) by trans-Arctic faunal exchange in both directions. Our revised date for Pacific-to-Atlantic mollusc migration removes the main piece of evidence for a previously hypothesized two-stage trans-Arctic migration event lasting over a million years.

Re-evaluation of trans-Arctic marine faunal exchange, the record of transgressions from sedimentary successions adjacent to the Bering Strait gateway, eustatic sea-level, and tectonic motions of Bering Strait suggests that the Bering Strait gateway was predominantly open for all of the Pliocene and the early Pleistocene (before 5 to 2 Ma). More frequent and longer lasting closures during glacial periods began at c. 1.6 Ma, and the gateway was predominantly closed after 1 Ma. Thus, we see no compelling evidence for previously proposed simple, direct links between the first opening of the Bering Strait oceanic gateway (c. 5.5 Ma), development of sea-ice in the Iceland Sea and initiation of the EGC (c. 4.5 Ma), and iNHG as signaled by major icesheet expansions in Greenland and Iceland (c. 3.5–2.7 Ma).

The topmost sediments in the Barmur Group and the lowermost lava in the Höskuldsvík Group lie within magnetic subchron C3n.2n at 4.631–4.493 Ma (Early Pliocene), older than all previous age models. The Barmur Group does not record the mid-Piacenzian Warm Period. There is no significant hiatus at the boundary between the Barmur and Höskuldsvík Groups. The upper Barmur Group (*Serripes* biozone) records each and every magnetic subchron down to the Skeifá lava, giving confidence in our new age model. Our revised, significantly older age for the Barmur Group implies that the upper Tjörnes succession accumulated over a correspondingly longer time period, which might be accommodated by longer durations for the Höskuldsvík and/or Breiðavík Groups, and/or a significant hiatus between these groups. Improving the age model for the upper Tjörnes sequence is an important aspiration because the lower Breiðavík Group provides the only onshore exposure record of the onset of lowland glaciation in Iceland.

The Kaldakvísl Group immediately underlying the Barmur Group has an Ar-Ar age of 9.8 ± 0.1 Ma (late Miocene). A lava at Grænhöfði previously thought to be within the Barmur Group has an Ar-Ar age of 9.3 ± 0.1 Ma. This lava is actually a faulted sliver of Kaldakvísl Group basement. We estimate that the age of the base of the Barmur Group is 6.0 ± 0.6 Ma (terminal Miocene) by linear extrapolation of the sedimentation rate determined for the upper Barmur Group. This age agrees with age-diagnostic dinoflagellate cysts and foraminifera, and also with the age of formation of the Tjörnes Basin.

Data Availability Statement

Data sets related to this article can be found at Hall et al. (2022), available via the Zenodo open data repository.

References

Albertsson, K. J. (1976). K/Ar ages of Pliocene-Pleistocene glaciations in Iceland with special reference to the Tjornes sequence, northern Iceland. (PhD). University of Cambridge.

Albertsson, K. J. (1978). Um aldur jarðlaga á Tjörnesi (in Icelandic). Náttúrufræðingurinn, 48, 1-8.

Aronson, J. L., & Sæmundsson, K. (1975). Relatively old basalts from structurally high areas in central Iceland. Earth and Planetary Science Letters, 28(1), 83–97. https://doi.org/10.1016/0012-821x(75)90077-1

Bartoli, G., Hönisch, B., & Zeebe, R. E. (2011). Atmospheric CO₂ decline during the Pliocene intensification of northern hemisphere glaciations. *Paleoceanography*, 26(42), PA4213. https://doi.org/10.1029/2010PA002055

Bárðarson, G. (1925). A stratigraphical survey of the Pliocene deposits at Tjörnes, in northern Iceland (Vol. 4). Det Kongelige Danske Videnskabernes Selskab, Biologiske Meddelelser.

Beets, C. J. (1992). Calibration of late Cenozoic marine strontium isotope variations and its chronostratigraphic and geochemical applications. (PhD). Vrije Universiteit Amsterdam.

Bennett, M. R., Bustos, D., Pigati, J. S., Springer, K. B., Urban, T. M., Holliday, V. T., et al. (2021). Evidence of humans in North America during the last glacial maximum. *Science*, 373(6562), 1528–1531. https://doi.org/10.1126/science.abg7586

This work was supported by the Natural Environment Research Council (NERC Grant NE/M00578X/1). We thank the NERC Isotope Geosciences Facilities Steering Committee for awarding funding to undertake rock preparation and 40Ar/39Ar dating at the SUERC Argon Isotope Facility (NIGL Grant IP-1855-1118). We are also grateful to Rhiannon Rees for facilitating the preparation of fused beads for whole rock geochemistry at the School of Ocean and Earth Science, University of Southampton (see Supporting Information S1). We thank Jón Eiríksson for ongoing advice and discussion regarding his new stratigraphic scheme for the Tiörnes succession, and two anonymous reviewers for helpful comments on the original manuscript.

25724525, 2023, 4, Downloaded

- Berggren, W. A., Kent, D. V., Swisher, C. C., III, Aubry, M.-P., Berggren, W. A., Kent, D. V., et al. (1995). A revised Cenozoic geochronology and Chronostratigraphy geochronology, time scales and global stratigraphic correlation (Vol. 54). SEPM Society for Sedimentary Geology. Brierley, C. M., & Fedorov, A. V. (2016). Comparing the impacts of Miocene-Pliocene changes in inter-ocean gateways on climate: Central Amer-
- ican Seaway, Bering Strait, and Indonesia. *Earth and Planetary Science Letters*, 444, 116–130. https://doi.org/10.1016/j.epsl.2016.03.010 Brigham-Grette, J., & Carter, L. D. (1992). Pliocene marine transgressions of northern Alaska: Circumarctic correlations and paleoclimatic
- interpretations. Arctic, 45(1), 74–89. https://doi.org/10.14430/arctic1375
 Buchardt, B., & Simonarson, L. A. (2003). Isotope palaeotemperatures from the Tjörnes beds in Iceland: Evidence of Pliocene cooling. Palaeogeography, Palaeoclimatology, Palaeoecology, 189(1–2), 71–95. https://doi.org/10.1016/s0031-0182(02)00594-1
- Campbell, L. D. (1993). Pliocene mollusks from the Yorktown and Chowan River formations in Virginia (Vol. 127, pp. 1–259). Virginia Division of Mineral Resources Publication.
- Camps, P., Singer, B. S., Carvallo, C., Goguitchaichvili, A., Fanjat, G., & Allen, B. (2011). The Kamikatsura event and the Matuyama–Brunhes reversal recorded in lavas from Tjörnes Peninsula, northern Iceland. *Earth and Planetary Science Letters*, 310(1), 33–44. https://doi. org/10.1016/j.epsl.2011.07.026
- Carter, L. D., Brigham-Grette, J., & Hopkins, D. M. (1986). Late Cenozoic marine transgressions of the Alaskan Arctic coastal plain. In J. A. Heginbottom & J.-S. Vincent (Eds.), Correlation of quaternary deposits and events around the margin of the Beaufort Sea: Contribution from a joint Canadian-American workshop (pp. 21–26). Geological Survey of Canada.
- Carter, L. D., Brigham-Grette, J., Marincovich, L., Jr., Pease, V. L., & Hillhouse, J. W. (1986). Late Cenozoic Arctic Ocean sea ice and terrestrial paleoclimate. *Geology*, 14(8), 675–678. https://doi.org/10.1130/0091-7613(1986)14<675:Lcaosi>2.0.Co;2
- Channell, J. E. T., Smelror, M., Jansen, E., Higgins, S. M., Lehman, B., Eidvin, T., & Solheim, A. (1999). Age models for glacial fan deposits off East Greenland and Svalbard (sites 986 and 987). In Proceedings of the ocean drilling program, scientific results, North Atlantic-Arctic gateways II; covering Leg 162 of the cruises of the drilling vessel JOIDES resolution (Vol. 162, p. 149). https://doi.org/10.2973/odp.proc. sr.162.008.1999
- Clotten, C., Stein, R., Fahl, K., Schreck, M., Risebrobakken, B., & De Schepper, S. (2019). On the causes of Arctic sea ice in the warm Early Pliocene. *Scientific Reports*, 9(1), 989. https://doi.org/10.1038/s41598-018-37047-y
- Cronin, T. M. (1991). Late Neogene marine Ostracoda from Tjörnes, Iceland. Journal of Paleontology, 65(5), 767–794. https://doi.org/10.1017/S002233600003777X
- De Schepper, S., Groeneveld, J., Naafs, B. D., Van Renterghem, C., Hennissen, J., Head, M. J., et al. (2013). Northern hemisphere glaciation during the globally warm early Late Pliocene. PLoS One, 8(12), e81508. https://doi.org/10.1371/journal.pone.0081508
- De Schepper, S., Head, M. J., & Louwye, S. (2009). Pliocene dinoflagellate cyst stratigraphy, palaeoecology and sequence stratigraphy of the Tunnel-Canal Dock, Belgium. *Geological Magazine*, 146(1), 92–112. https://doi.org/10.1017/s0016756808005438
- De Schepper, S., Schreck, M., Beck, K. M., Matthiessen, J., Fahl, K., & Mangerud, G. (2015). Early Pliocene onset of modern Nordic Seas circulation related to ocean gateway changes. *Nature Communications*, 6(1), 8659. https://doi.org/10.1038/ncomms9659
- Doell, R. R. (1972). Palaeomagnetic studies of Icelandic lava flows. *Geophysical Journal International*, 26(5), 459–479. https://doi.org/10.1111/ j.1365-246X.1972.tb05763.x
- Doré, A. G., & Lundin, E. R. (1996). Cenozoic compressional structures on the NE Atlantic margin; nature, origin and potential significance for hydrocarbon exploration. *Petroleum Geoscience*, 2(4), 299–311. https://doi.org/10.1144/petgeo.2.4.299
- Dowsett, H. J., Barron, J., & Poore, R. (1996). Middle Pliocene sea surface temperatures: A global reconstruction. *Marine Micropaleontology*, 27(1), 13–25. https://doi.org/10.1016/0377-8398(95)00050-X
- Dowsett, H. J., Thompson, R., Barron, J., Cronin, T., Fleming, F., Ishman, S., et al. (1994). Joint investigations of the Middle Pliocene climate I: PRISM paleoenvironmental reconstructions. *Global and Planetary Change*, 9(3), 169–195. https://doi.org/10.1016/0921-8181(94)90015-9
- Dowsett, H. J., Barron, J. A., Poore, R. Z., Thompson, R. S., Cronin, T. M., Ishman, S. E., & Willard, D. A. (1999). Middle Pliocene paleoenvironmental reconstruction (pp. 99–535). PRISM2. Retrieved from http://pubs.er.usgs.gov/publication/ofr99535
- Dowsett, H. J., Dolan, A. M., Rowley, D., Pound, M., Salzmann, U., Robinson, M. M., et al. (2016). The PRISM4 (mid-Piacenzian) palaeoenvironmental reconstruction. *Climate of the Past Discussions*, 1–39. https://doi.org/10.5194/cp-2016-33
- Dowsett, H. J., & Poore, R. Z. (1991). Pliocene sea surface temperatures of the North Atlantic Ocean at 3.0 Ma. Quaternary Science Reviews, 10(2), 189–204. https://doi.org/10.1016/0277-3791(91)90018-P
- Dowsett, H. J., Robinson, M. M., Foley, K. M., & Herbert, T. D. (2021). The Yorktown formation: Improved stratigraphy, chronology, and paleoclimate interpretations from the U.S. Mid-Atlantic coastal plain. *Geosciences*, 11(12), 486. https://doi.org/10.3390/geosciences11120486
- Dowsett, H. J., Robinson, M. M., Haywood, A. M., Salzmann, U., Hill, D. J., Sohl, L. E., et al. (2010). The PRISM3D paleoenvironmental reconstruction. Stratigraphy, 7(2–3), 123–139.
- Driscoll, N. W., & Haug, G. H. (1998). A short circuit in thermohaline circulation: A cause for northern hemisphere glaciation? Science, 282(5388), 436–438. https://doi.org/10.1126/science.282.5388.436
- Durham, J. W., & MacNeil, F. S. (1967). Cenozoic migrations of marine invertebrates through the Bering Strait region. In D. M. Hopkins (Ed.), *The Bering land bridge* (pp. 326–349). Stanford University Press.
- Einarsson, T., Hopkins, D. M., & Doell, R. R. (1967). The stratigraphy of Tjörnes, northern Iceland, and the history of the Bering land bridge. In D. M. Hopkins (Ed.), *The Bering land bridge* (pp. 313–325). Stanford University Press.
- Eiríksson, J. (1980). Tjornes, north Iceland: A bibliographical review of the geological research history. *Jökull Journal*, 30, 1–20. https://doi.org/10.33799/jokull1980.30.001
- Eiríksson, J. (1981). Lithostratigraphy of the upper Tjornes sequence, north Iceland: The Breiðavík Group. Acta Naturalia Islandica, 29, 1–37. Eiríksson, J. (1985). Facies analysis of the Breiðavík group sediments on Tjörnes, north Iceland. Acta Naturalia Islandica, 31, 1–56.
- Eiríksson, J., & Geirsdóttir, Á. (1991). A record of Pliocene and Pleistocene glaciations and climatic changes in the North Atlantic based on variations in volcanic and sedimentary facies in Iceland. *Marine Geology*, 101(1–4), 147–159. https://doi.org/10.1016/0025-3227(91)90068-f
- Eiríksson, J., Gudmundsson, A. I., Kristjánsson, L., & Gunnarsson, K. (1990). Palaeomagnetism of Pliocene-Pleistocene sediments and lava flows on Tjörnes and Flatey, north Iceland. Boreas, 19(1), 39–55. https://doi.org/10.1111/j.1502-3885.1990.tb00420.x
- Eiríksson, J., Guðmundsson, A. I., Símonarson, L. A., & Knudsen, K. L. (2021). Lithostratigraphy of the upper part of the Tjörnes sequence in Furuvík, Breiðavík, Öxarfjörður, and central Tjörnes Mountains, north Iceland. In J. Eiríksson & L. A. Símonarson (Eds.), Pacific - Atlantic mollusc migration: Pliocene inter-ocean gateway archives on Tjörnes, north Iceland (pp. 567–665). Springer International Publishing.
- Eiríksson, J., Knudsen, K. L., & Símonarson, L. A. (2021). Reconstructing the Paleoenvironments of the quaternary Tjörnes Basin, north Iceland. In J. Eiríksson & L. A. Símonarson (Eds.), Pacific - Atlantic mollusc migration: Pliocene inter-ocean gateway archives on Tjörnes, north Iceland (pp. 803–839). Springer International Publishing.
- Eiríksson, J., Knudsen, K. L., & Vilhjálmsson, M. (1992). An early pleistocene glacial-interglacial cycle in the Breiðavík group on Tjörnes, Iceland: Sedimentary facies, foraminifera, and molluscs. *Quaternary Science Reviews*, 11(7–8), 733–757. https://doi.org/10.1016/0277-3791(92)90080-r

- Eiríksson, J., & Símonarson, L. A. (2021). A review of the research history of the Tjörnes sequence, north Iceland. In J. Eiríksson & L. A. Símonarson (Eds.), Pacific - Atlantic mollusc migration: Pliocene inter-ocean gateway archives on Tjörnes, north Iceland (pp. 57–91). Springer International Publishing.
- Eiríksson, J., Símonarson, L. A., & Knudsen, K. L. (2021a). An age model for the Miocene to pleistocene Tjörnes sequence, north Iceland. In J. Eiríksson & L. A. Símonarson (Eds.), Pacific - Atlantic mollusc migration: Pliocene inter-ocean gateway archives on Tjörnes, north Iceland (pp. 213–236). Springer International Publishing.
- Eiríksson, J., Símonarson, L. A., & Knudsen, K. L. (2021b). Lithostratigraphy of the Tjörnes sequence in Barmur and Höskuldsvík on the west coast of Tjörnes, north Iceland. In J. Eiríksson & L. A. Símonarson (Eds.), Pacific - Atlantic mollusc migration: Pliocene inter-ocean gateway archives on Tjörnes, north Iceland (pp. 93–212). Springer International Publishing.
- Elliott, J., & Freymueller, J. T. (2020). A block model of present-day kinematics of Alaska and Western Canada. *Journal of Geophysical Research:* Solid Earth, 125(7), e2019JB018378. https://doi.org/10.1029/2019JB018378
- Field, D. J., Boessenecker, R., Racicot, R. A., Ásbjörnsdóttir, L., Jónasson, K., Hsiang, A. Y., et al. (2017). The oldest marine vertebrate fossil from the volcanic island of Iceland: A partial right whale skull from the high latitude Pliocene Tjörnes Formation. *Palaeontology*, 60(2), 141–148. https://doi.org/10.1111/pala.12275
- Fiorillo, A. R., Blodgett, R. B., & Stanley, G. D., Jr. (2008). Dinosaurs of Alaska: Implications for the cretaceous origin of Beringia. In The terrane puzzle: New perspectives on paleontology and stratigraphy from the North American Cordillera (Vol. 442). Geological Society of America.
- Garcia, S., Arnaud, N. O., Angelier, J., Bergerat, F., & Homberg, C. (2003). Rift jump process in Northern Iceland since 10 Ma from ⁴⁰Ar/³⁹Ar geochronology. *Earth and Planetary Science Letters*, 214(3–4), 529–544. https://doi.org/10.1016/s0012-821x(03)00400-x
- Geirsdóttir, Á., & Eiríksson, J. (1994). Growth of an intermittent ice sheet in Iceland during the late Pliocene and early Pleistocene. Quaternary Research, 42(2), 115–130. https://doi.org/10.1006/qres.1994.1061
- Gladenkov, A. Y., & Gladenkov, Y. B. (2004). Onset of connections between the Pacific and Arctic Oceans through the Bering Strait in the Neogene. Stratigraphy and Geological Correlation, 12(2), 175–187.
- Gladenkov, A. Y., Oleinik, A. E., Marincovich, L., & Barinov, K. B. (2002). A refined age for the earliest opening of Bering Strait. Palaeogeography, Palaeoclimatology, Palaeoecology, 183(3), 321–328. https://doi.org/10.1016/S0031-0182(02)00249-3
- Gladenkov, Y. B. (1981). Marine Plio-Pleistocene of Iceland and problems of its correlation. *Quaternary Research*, 15(1), 18–23. https://doi.org/10.1016/0033-5894(81)90111-3
- Gladenkov, Y. B., & Gurari, G. Z. (1976). Paleomagnetic characteristics of the Pliocene-Pleistocene in Iceland (Tjörnes Peninsula) (pp. 1173–1175). Doklady Academic Nauk.
- Goebel, T., Waters, M. R., & O'Rourke, D. H. (2008). The late pleistocene dispersal of modern humans in the Americas. *Science*, 319(5869), 1497–1502. https://doi.org/10.1126/science.1153569
- Gusev, E. A., Andreeva, I. A., Anikina, N. Y., Bondarenko, S. A., Derevyanko, L. G., Iosifidi, A. G., et al. (2009). Stratigraphy of Late Cenozoic sediments of the western Chukchi Sea: New results from shallow drilling and seismic-reflection profiling. *Global and Planetary Change*, 68(1), 115–131. https://doi.org/10.1016/j.gloplacha.2009.03.025
- Hall, J. R., Allison, M. S., Papadopoulos, M. T., Barfod, D. N., & Jones, S. M. (2022). ⁴⁰Ar/³⁹Ar dating of the Barmur Group (Tjörnes beds), northern Iceland. [Dataset]. Zenodo. https://doi.org/10.5281/zenodo.7016669
- Haug, G. H., & Tiedemann, R. (1998). Effect of the formation of the Isthmus of Panama on Atlantic Ocean thermohaline circulation. Nature, 393(6686), 673–676. https://doi.org/10.1038/31447
- Haywood, A. M., Dowsett, H. J., Dolan, A. M., Rowley, D., Abe-Ouchi, A., Otto-Bliesner, B., et al. (2016). The Pliocene model intercomparison Project (PlioMIP) phase 2: Scientific objectives and experimental design. *Climate of the Past*, 12(3), 663–675. https://doi.org/10.5194/ cp-12-663-2016
- Haywood, A. M., Tindall, J. C., Dowsett, H. J., Dolan, A. M., Foley, K. M., Hunter, S. J., et al. (2020). The Pliocene model intercomparison Project phase 2: Large-scale climate features and climate sensitivity. *Climate of the Past*, 16(6), 2095–2123. https://doi.org/10.5194/cp-16-2095-2020
- Hopkins, D. M. (1959). Cenozoic history of the Bering land bridge. Science, 129(3362), 1519–1528. https://doi.org/10.1126/science.129.3362.1519
 Hopkins, D. M. (1967). Quaternary marine transgressions in Alaska. In D. M. Hopkins (Ed.), The Bering land bridge (pp. 47–90). Stanford University Press.
- Hopkins, D. M. (1972). The paleogeography and climatic history of Beringia during late Cenozoic time. Inter-Nord: Revue internationale d'études arctiques et nordiques, (12), 121–150.

Horikawa, K., Martin, E. E., Basak, C., Onodera, J., Seki, O., Sakamoto, T., et al. (2015). Pliocene cooling enhanced by flow of low-salinity Bering Sea water to the Arctic Ocean. *Nature Communications*, 6(1), 7587. https://doi.org/10.1038/ncomms8587

- Hospers, J. (1953). Palaeomagnetic studies of Icelandic rocks. (PhD).
- Hu, A., Meehl, G. A., Otto-Bliesner, B. L., Waelbroeck, C., Han, W., Loutre, M.-F., et al. (2010). Influence of Bering Strait flow and North Atlantic circulation on glacial sea-level changes. *Nature Geoscience*, 3(2), 118–121. https://doi.org/10.1038/ngeo729
- Jones, S. M., White, N., & Maclennan, J. (2002). V-shaped ridges around Iceland: Implications for spatial and temporal patterns of mantle convection. Geochemistry, Geophysics, Geosystems, 3(10), 1–23. https://doi.org/10.1029/2002gc000361
- Kaufman, D. S., & Brigham-Grette, J. (1993). Aminostratigraphic correlations and paleotemperature implications, Pliocene-Pleistocene high-sea-level deposits, northwestern Alaska. *Quaternary Science Reviews*, 12(1), 21–33. https://doi.org/10.1016/0277-3791(93)90046-O
- Knies, J., Cabedo-Sanz, P., Belt, S. T., Baranwal, S., Fietz, S., & Rosell-Melé, A. (2014). The emergence of modern sea ice cover in the Arctic Ocean. Nature Communications, 5(1), 5608. https://doi.org/10.1038/ncomms6608
- Knudsen, K. L., Eiríksson, J., & Símonarson, L. A. (2021). Foraminifera in the Pliocene Barmur group on Tjörnes, north Iceland. In J. Eiríksson & L. A. Símonarson (Eds.), Pacific - Atlantic mollusc migration: Pliocene inter-ocean gateway archives on Tjörnes, north Iceland (pp. 443–524). Springer International Publishing.
- Kristjánsson, L. (2004). A reconnaissance study of paleomagnetic directions in the Tjörnes beds, Northern Iceland. *Jökull Journal*, 54(1), 57–63. https://doi.org/10.33799/jokull2004.54.057
- Lagoe, M. B. (1992). Neogene paleoclimate of the Gulf of Alaska and regional correlations to the Bering Sea region. In *Paper presented at the* 1992 proceeding International conference on the Arctic margins.
- Lambeck, K. (2004). Sea-level change through the last glacial cycle: Geophysical, glaciological and palaeogeographic consequences. Comptes Rendus Geoscience, 336(7), 677–689. https://doi.org/10.1016/j.crte.2003.12.017
- Lisiecki, L. E., & Raymo, M. E. (2005). A Pliocene-Pleistocene stack of 57 globally distributed benthic δ¹⁸O records. *Paleoceanography*, 20(1), PA1003. https://doi.org/10.1029/2004pa001071
- Lunt, D. J., Foster, G. L., Haywood, A. M., & Stone, E. J. (2008). Late Pliocene Greenland glaciation controlled by a decline in atmospheric CO₂ levels. *Nature*, 454(7208), 1102–1105. https://doi.org/10.1038/nature07223

25724525, 2023, 4, Downloaded

- MacNeil, F. S. (1965). Evolution and distribution of the genus Mya, and Tertiary migrations of Mollusca (483G). Retrieved from http://pubs. er.usgs.gov/publication/pp483G
- Maier-Reimer, E., Mikolajewicz, U., & Crowley, T. (1990). Ocean general circulation model sensitivity experiment with an open central American Isthmus. *Paleoceanography*, 5(3), 349–366. https://doi.org/10.1029/PA005i003p00349
- Marincovich, L., & Gladenkov, A. Y. (1999). Evidence for an early opening of the Bering Strait. *Nature*, 397(6715), 149–151. https://doi.org/10.1038/16446
- Marincovich, L., & Gladenkov, A. Y. (2001). New evidence for the age of Bering Strait. Quaternary Science Reviews, 20(1), 329–335. https:// doi.org/10.1016/S0277-3791(00)00113-X
- Marincovich, L., Jr. (2000). Central American paleogeography controlled Pliocene Arctic Ocean molluscan migrations. *Geology*, 28(6), 551. https://doi.org/10.1130/0091-7613(2000)28<551:capcpa>2.0.co;2
- Marincovich, L., Jr., Brouwers, E. M., Hopkins, D. M., McKenna, M. C., Grantz, A., Johnson, L., & Sweeney, J. F. (1990). Late Mesozoic and Cenozoic paleogeographic and paleoclimatic history of the Arctic Ocean Basin, based on shallow-water marine faunas and terrestrial vertebrates. In *The Arctic Ocean region*. Geological Society of America.
- Mark, D. F., Barfod, D., Stuart, F. M., & Imlach, J. (2009). The ARGUS multicollector noble gas mass spectrometer: Performance for ⁴⁰Arl³⁹Ar geochronology. *Geochemistry, Geophysics, Geosystems*, 10(10), Q0AA02. https://doi.org/10.1029/2009GC002643
- Maslin, M. A., Li, X. S., Loutre, M. F., & Berger, A. (1998). The contribution of orbital forcing to the progressive intensification of northern hemisphere glaciation. *Quaternary Science Reviews*, 17(4–5), 411–426. https://doi.org/10.1016/s0277-3791(97)00047-4
- McDougall, I., & Harrison, T. M. (1999). Geochronology and Thermochronology by the ⁴⁰Ar/³⁹Ar method (2nd ed., p. 284). Oxford University Press.
- Meijer, T. (1993). Stratigraphical notes on Macoma (Bivalvia) in the southern part of the North Sea Basin and some remarks on the arrival of Pacific species. Scripta Geologica Special Issues, 2(15), 297–312.
- Miller, K. G., Kominz, M. A., Browning, J. V., Wright, J. D., Mountain, G. S., Katz, M. E., et al. (2005). The Phanerozoic record of global sea-level change. Science, 310(5752), 1293–1298. https://doi.org/10.1126/science.1116412
- Moreno-Mayar, J. V., Potter, B. A., Vinner, L., Steinrücken, M., Rasmussen, S., Terhorst, J., et al. (2018). Terminal Pleistocene Alaskan genome reveals first founding population of Native Americans. *Nature*, 553(7687), 203–207. https://doi.org/10.1038/nature25173
- Niespolo, E. M., Rutte, D., Deino, A. L., & Renne, P. R. (2017). Intercalibration and age of the Alder Creek sanidine ⁴⁰Ar/³⁹Ar standard. *Quaternary Geochronology*, 39, 205–213. https://doi.org/10.1016/j.quageo.2016.09.004
- Ogg, J. G. (2020). Chapter 5 Geomagnetic polarity time scale. In F. M. Gradstein, J. G. Ogg, M. D. Schmitz, & G. M. Ogg (Eds.), Geologic time Scale 2020 (pp. 159–192). Elsevier.
- O'Regan, M., John, K. S., Moran, K., Backman, J., King, J., Haley, B. A., et al. (2010). Plio-Pleistocene trends in ice rafted debris on the Lomonosov Ridge. *Quaternary International*, 219(1), 168–176. https://doi.org/10.1016/j.quaint.2009.08.010
- Parnell-Turner, R., White, N. J., McCave, I. N., Henstock, T. J., Murton, B., & Jones, S. M. (2015). Architecture of North Atlantic contourite drifts modified by transient circulation of the Icelandic mantle plume. *Geochemistry, Geophysics, Geosystems*, 16(10), 3414–3435. https:// doi.org/10.1002/2015gc005947
- Pflug, H. D. (1959). Sporenbilder aus Island und ihre stratigraphische Deutung (in German). Neues Jahrbuch für Geologie und Paläontologie, 107(2), 141–172.
- Poore, H. R., Samworth, R., White, N. J., Jones, S. M., & McCave, I. N. (2006). Neogene overflow of northern component water at the Greenland-Scotland ridge. *Geochemistry, Geophysics, Geosystems*, 7(6), Q06010. https://doi.org/10.1029/2005GC001085
- Poulsen, N. E., Manum, S. B., Williams, G. L., & Ellegaard, M. (1996). Tertiary dinoflagellate biostratigraphy of sites 907, 908, and 909 in the Norwegian-Greenland Sea. In Proceedings of the ocean drilling program; scientific results, North Atlantic-Arctic gateways I; covering Leg 151 of the cruises of the drilling vessel JOIDES resolution (Vol. 151, p. 255). https://doi.org/10.2973/odp.proc.sr.151.110.1996
- Raffi, I., Wade, B. S., Pälike, H., Beu, A. G., Cooper, R., Crundwell, M. P., et al. (2020). Chapter 29 The Neogene period. In F. M. Gradstein, J. G. Ogg, M. D. Schmitz, & G. M. Ogg (Eds.), *Geologic time Scale 2020* (pp. 1141–1215). Elsevier.
- Redfield, T. F., Scholl, D. W., Fitzgerald, P. G., & Beck, M. E., Jr. (2007). Escape tectonics and the extrusion of Alaska: Past, present, and future. *Geology*, 35(11), 1039–1042. https://doi.org/10.1130/g23799a.1
- Renne, P. R., Mundil, R., Balco, G., Min, K., & Ludwig, K. R. (2010). Joint determination of ⁴⁰K decay constants and ⁴⁰Ar*/⁴⁰K for the Fish Canyon sanidine standard, and improved accuracy for ⁴⁰Ar/³⁹Ar geochronology. *Geochimica et Cosmochimica Acta*, 74(18), 5349–5367. https://doi.org/10.1016/j.gca.2010.06.017
- Robinson, M. M., Dowsett, H. J., Foley, K. M., & Riesselman, C. R. (2018). PRISM marine sites The history of PRISM sea surface temperature estimation. Retrieved from http://pubs.er.usgs.gov/publication/ofr20181148
- Sæmundsson, K. (1974). Evolution of the axial rifting zone in northern Iceland and the Tjörnes fracture zone. *Geological Society of America Bulletin*, 85(4), 495–504. https://doi.org/10.1130/0016-7606(1974)85<495:EOTARZ>2.0.CO2
- Sarnthein, M., Bartoli, G., Prange, M., Schmittner, A., Schneider, B., Weinelt, M., et al. (2009). Mid-Pliocene shifts in ocean overturning circulation and the onset of Quaternary-style climates. *Climate of the Past*, 5(2), 269–283. https://doi.org/10.5194/cp-5-269-2009
- Scholl, D., Stevenson, A., Mueller, S., Geist, E., Engebretson, D., & Vallier, T. (1992). Exploring the notion that southeast-Asian-type escape tectonics and trench clogging are involved in regional-scale deformation of Alaska and the formation of the Aleutian-Bering Sea region. In Paper presented at the southeast Asia structure, tectonics and magmatism, proc. Geod. Res. Inst. Symp.
- Schwarzbach, M., & Pflug, H. D. (1957). Beiträge zur Klimageschichte Islands 6 Das Klima des Jüngeren Tertiärs in Island (in German). Neues Jahrbuch für Geologie und Paläontologie Abhandlungen, 104, 279–298.
- Seki, O., Foster, G. L., Schmidt, D. N., Mackensen, A., Kawamura, K., & Pancost, R. D. (2010). Alkenone and boron-based Pliocene pCO₂ records. *Earth and Planetary Science Letters*, 292(1–2), 201–211. https://doi.org/10.1016/j.epsl.2010.01.037
- Sigmundsson, F., Einarsson, P., Hjartardóttir, Á. R., Drouin, V., Jónsdóttir, K., Árnadóttir, T., et al. (2020). Geodynamics of Iceland and the signatures of plate spreading. Journal of Volcanology and Geothermal Research, 391, 106436. https://doi.org/10.1016/j.jvolgeores.2018.08.014
- Símonarson, L. A., & Eiríksson, J. (2008). Tjörnes-Pliocene and Pleistocene sediments and faunas. Jökull Journal, 58(1), 331–342. https://doi. org/10.33799/jokull2008.58.331
- Símonarson, L. A., & Eiríksson, J. (2021a). Migration of Pacific marine mollusc fauna into the North Atlantic Across the Arctic Ocean in Pliocene and early pleistocene time. In J. Eiríksson & L. A. Símonarson (Eds.), Pacific - Atlantic mollusc migration: Pliocene inter-ocean gateway archives on Tjörnes, north Iceland (pp. 841–867). Springer International Publishing.
- Símonarson, L. A., & Eiríksson, J. (2021b). Systematic overview of the Pliocene molluscs and Barnacles of the Barmur group on Tjörnes, north Iceland. In J. Eiríksson & L. A. Símonarson (Eds.), Pacific - Atlantic mollusc migration: Pliocene inter-ocean gateway archives on Tjörnes, north Iceland (pp. 237–441). Springer International Publishing.



- Símonarson, L. A., Knudsen, K. L., & Eiríksson, J. (2021). Reconstructing the Paleoenvironments of the Pliocene Barmur group in the Tjörnes Basin, north Iceland. In J. Eiríksson & L. A. Símonarson (Eds.), Pacific - Atlantic mollusc migration: Pliocene inter-ocean gateway archives on Tjörnes, north Iceland (pp. 525–565). Springer International Publishing.
- Stein, R., Fahl, K., Schreck, M., Knorr, G., Niessen, F., Forwick, M., et al. (2016). Evidence for ice-free summers in the late Miocene central Arctic Ocean. Nature Communications, 7(1), 11148. https://doi.org/10.1038/ncomms11148
- Strauch, F. (1963). Zur Geologie von Tjörnes (Nordisland) (in German) (Vol. 7). Sonderveröffentlichungen des Geologischen Instituts der Universität Köln.
- Strauch, F. (1972a). Phylogenese, Adaptation und Migration einiger nordischer mariner Molluskengenera (Neptunea, Panomya, Cyrtodaria und Mya) (in German). Abhandlungen der Senckenbergischen Naturforschenden Gesellschaft, 531, 1–210.
- Strauch, F. (1972b). Zum Klima des nordatlantisch-skandischen Raumes im jüngeren Känozoikum (in German). Zeitschrift der Deutschen Geologischen Gesellschaft, 123(1), 163–177. https://doi.org/10.1127/zdgg/123/1972/163
- Straume, E. O., Gaina, C., Medvedev, S., & Nisancioglu, K. H. (2020). Global Cenozoic Paleobathymetry with a focus on the Northern Hemisphere oceanic gateways. *Gondwana Research*, 86, 126–143. https://doi.org/10.1016/j.gr.2020.05.011
- Thoroddsen, T. (1902). Islandske Fjorde og Bugter (in Danish). Geografisk Tidsskrift, 16, 58-82.
- Torsvik, T. H., Carlos, D., Mosar, J., Cocks, L. R. M., & Malme, T. (2002). Global reconstructions and North Atlantic paleogeography 440 Ma to recent. In BATLAS - Mid Norway plate reconstruction atlas with global and Atlantic perspectives (Vol. 18, p. 39). Geological Survey of Norway.
- Verhoeven, K., & Louwye, S. (2013). Palaeoenvironmental reconstruction and biostratigraphy with marine palynomorphs of the Plio–Pleistocene in Tjörnes, Northern Iceland. *Palaeogeography, Palaeoclimatology, Palaeoecology*, 376, 224–243. https://doi.org/10.1016/j.palaeo.2013.03.002
- Verhoeven, K., Louwye, S., & Eiríksson, J. (2013). Plio-Pleistocene landscape and vegetation reconstruction of the coastal area of the Tjörnes Peninsula, Northern Iceland. *Boreas*, 42(1), 108–122. https://doi.org/10.1111/j.1502-3885.2012.00279.x
- Verhoeven, K., Louwye, S., Eiríksson, J., & De Schepper, S. (2011). A new age model for the Pliocene–Pleistocene Tjörnes section on Iceland: Its implication for the timing of North Atlantic–Pacific palaeoceanographic pathways. *Palaeogeography, Palaeoclimatology, Palaeoecology,* 309(1–2), 33–52. https://doi.org/10.1016/j.palaeo.2011.04.001
- Walters, R. L., Jones, S. M., & Maclennan, J. (2013). Renewed melting at the abandoned Húnafloí Rift, northern Iceland, caused by plume pulsing. Earth and Planetary Science Letters, 377–378, 227–238. https://doi.org/10.1016/j.epsl.2013.06.040
- Willeit, M., Ganopolski, A., Calov, R., Robinson, A., & Maslin, M. (2015). The role of CO₂ decline for the onset of Northern Hemisphere glaciation. *Quaternary Science Reviews*, 119, 22–34. https://doi.org/10.1016/j.quascirev.2015.04.015
- Woodgate, R. A., & Aagaard, K. (2005). Revising the Bering Strait freshwater flux into the Arctic Ocean. Geophysical Research Letters, 32(2), L02602. https://doi.org/10.1029/2004GL021747
- Zachos, J., Pagani, M., Sloan, L., Thomas, E., & Billups, K. (2001). Trends, rhythms, and Aberrations in global climate 65 Ma to present. Science, 292(5517), 686–693. https://doi.org/10.1126/science.1059412

References From the Supporting Information

Le Bas, M. J., Le Maitre, R. W., Streckeisen, A., & Zanettin, B. (1986). A chemical classification of volcanic rocks based on the total Alkali-Silica diagram. Journal of Petrology, 27(3), 745–750. https://doi.org/10.1093/petrology/27.3.745